Reputation, Quality Choices and Optimal Product Discontinuation Policy

PORNTHEP BENYAAPIKUL[†] Department of Economics, University of Oxford Faculty of Economics, Thammasat University

> June 2007 Preliminary Draft

Abstract

This paper studies quality choice and optimal product discontinuation decisions when a firm's intended quality choice is difficult to control and information about actual quality is noisy. The market forms beliefs about the actual quality based on the observed sequence of noisy signals or product's reputation. It is shown that optimal product discontinuation policy exists. A firm discontinues its product when reputation or realised quality falls below a threshold. Incentives for providing higher quality arise from two channels: higher quality increases (i) consumers' valuation for the product conditional on its being continued and (ii) the product's chance of surviving the policy. The firm exerts higher quality and applies a more stringent reputation standard discontinuation policy when incentives from both channels are strengthened. This happens when either the firm cares more about the long-term relationship or the precision of the initial prior belief decreases. Signal informativeness has ambiguous effects on the optimal discontinuation policy as its effects on consumers' valuation are unclear. However, without discontinuation policy, higher informativeness unambiguously increases quality.

Key words: Consumer Learning, Optimal Stopping, Product Discontinuation. JEL classification: D21, D83, L14, L15

[†]Email: pornthep.benyaapikul@economics.ox.ac.uk. I am grateful to David Myatt and Sittisak Leelahanon for excellent suggestions and comments. Thank to Swarnali Ahmed, Pathomwat Chantarasap and Wanwiphang Manachotphong for many helpful discussions.

Character is like a tree and reputation like its shadow. The shadow is what we think of it; the tree is the real thing.

-Abraham Lincoln

1 Introduction

It has long been known that imperfect information can cause a wide range of market imperfections. One example that has been widely discussed is market failure when the product's quality is unobservable. This is particularly the problem where consumers are only able to evaluate the quality of the products after they have been bought and consumed. These products are known as "experience goods". Examples include durables such as cars or used cars, electronic appliances, nondurables such as wine, and services such as airlines or specialist professions. Usually, the problem occurs even in a market where the product's characteristics are observable or information on inherited quality is available to the market but a firm cannot perfectly control the intended quality then the true product's quality is unknown to the market and consumers' utility derived from the products requires a match between product characteristics or features and consumer's taste or preference. For instance, technology gadgets such as the mobile phone: information on its features (camera resolution, 3G compatibility, lightness of weight) is available, but all the features available on it may be rarely used by consumers and they may find these out only after purchase. Fashion or designer branded products are another example: the quality of the materials may be observable, but the consumer may not like them because they may not match the consumer's preference or taste. Many mechanisms have been proposed to solve this problem. One possible way is by repeated interactions, where the consumer frequently purchase the product and is able to observed its realised quality, although still not able to observe the firm's efforts in quality. In these settings, the consumer's experiences with a particular product provide imperfect information on a combination of the firm's efforts, a suiting of their tastes, and sheer luck. This collection of experiences forms an expectation of quality, and is the definition of reputation for this paper.

As the learning process continues, information on the quality of the product is gradually revealed to the market. Many producers, after observing the sequence of the product's quality reputation over some periods, decide to take the product out of the market and launch a new one before the quality of the old product is completely revealed. The market then has to start the learning process over again. These "discontinuation policies" typically are deterministic or fixed in advance, for example, cars where Japanese car manufacturers have a policy of launching a new car design approximately every 5 years while German manufacturers do this every 7 years, and designer clothes where a new fashion is introduced every season. However, some firms adopt a discontinuation policy that is not based on the length of time that the product has been on the market. Different firms may apply different criteria for discontinuing their products. Some products such as computer processors are discontinued because of the advancement of new technology. Some firms launch a new product in response to their competitors. For example, Sony was planning to launch its new generation game console Play Station 3 (PS3) in the year 2007. However, after its topof-the-line rival Microsoft's Xbox 360 was released in November 2005, Sony revised their PS3 launching plan to the spring of 2006. Nevertheless, because of a technical problem, the PS3 was finally released in November 2006. Some firms use their product's quality deteriorating reputation as a criterion for discontinuing. A good example is the series of battery recalls from many computer manufacturers such as Lenovo; this manufaturer announced a recall of 526,000 laptop batteries on September 2006, after a Lenovo ThinkPad T43 caught fire at the Los Angeles International Airport. Or the notebook battery recalls issued by Dell and Apple Computer in August 2006, after it was discovered that some batteries were overheating or exploding, causing fires even when the machines were turned off.¹ It is interesting to see how firms optimally determine their discontinuation policy when information about product quality is imperfect and how the policy changes when the informativeness of consumers' experiences changes.

This paper studies quality choice and product discontinuation problems under a moral hazard when information about product quality is imperfect but symmetric; information about the true product quality is unknown to both consumers and firm. The firm's intended quality choice is difficult to control and is subject to a random shock. The paper focuses in particular on product discontinuation policies when the firm uses quality reputation as a criterion of discontinuation.

The paper is organized as follows. Section 2 reviews the related literature. Section 3 sets up the basic model and discusses some results and implications. Section 4 introduces the problem of product discontinuation by considering the case where all products have a predetermined time in the market before being discontinued. Section 5 analyses the

¹http://news.zdnet.co.uk/hardware/0,1000000091,39283811,00.htm

discontinuation problem when the product stopping period is stochastic and shows that an optimal discontinuation policy exists. The optimal policy is then derived and comparative static analyses are carried out. Section 6 presents the conditions for optimal quality choice under optimal product discontinuation policy, and discusses how a product discontinuation policy can affect a firm's incentive to provide quality. Section 7 provides conclusions.

2 Literature Review

Reputation has been extensively studied in economics as a mechanism that could alleviate inefficiencies arising from imperfect information. The notions of reputation in economics can be categorized into two mechanisms (Cabral 2005). The first is where the firm's or agent's action is unobservable; a moral hazard problem. Reputation may arise either from repeated interactions between a firm and consumers or be an implicit contract between two parties. The second involves adverse selection problems where there are information asymmetries between the firm and the consumers. Quality is a characteristic or type of firms that is known exclusively to them. In this case, uninformed consumers have access to the history of past stage outcomes and form their poseterior beliefs on this regarding the firm's type. The informed firm can improve its long-run profit by acting in a way that will convince consumers to believe that it belongs to the better type.

For reputation mechanisms under moral hazard, the seminal work of Klien and Leffler (1981) suggests that reputation mechanisms can ensure high quality performance in circumstances where quality is not observable. In order to sustain a high reputation and produce high quality goods, the returns to the firm must be high and persuasive enough to offset a short-run incentive to deviate. The idea is explored more formally in Shapiro (1983), where different firms specialise in different quality products and entry and exit are incorporated. Reputation is defined as the consumer's expectation of quality, and it has been shown that reputation can work as an imperfect mechanism for quality assurance and that high quality products are sold at a premium above their cost. Both papers' analyses are based on the assumption that the experience from consuming the product delivers the perfect signal about quality. The consumers knows the quality completely and immediately after consumption of the product.² Rogerson (1983) considers the role of reputation in assuring quality where consumers are able to observe quality after purchase imperfectly

 $^{^{2}}$ Dybvig and Spatt (1982) and von Weizsäcker (1980) apply a similar idea of a perfect signal of quality, but consumers become aware of this with a lag.

and probabilistically.

An idea similar to that of Klien and Leffler (1981) appears in labour economic literature; Fama (1980) explains situations where the labour market does not know the agent's true ability or talent but learns this from observing past performances. However, agents are disciplined by implicit contracts or their career concerns; the market pays according to its belief about an agent's ability. In other words, an agent's compensation depends upon the market's belief of his ability and reputation refers to this belief. This idea is formalised by Holmström (1982). In this celebrated paper, information about an agent's ability is imperfect but symmetric, i.e. it is unknown to both agent and employers. Quality produced by the agent is a combination of the agent's talent, effort and luck and is a noisy signal to actual ability. Holmström (1982) shows that, at the beginning of a relationship where an agent's talent is unknown, the agent exerts high effort to produce high quality. As time progresses, the market updates the belief about the agent's ability and, finally, this will be completely realised. Efforts, therefore, decrease over time. Holmström (1982)'s reasoning shows that no equilibrium in which a single firm commits to a high quality product exists.

As noted by Mailath and Samuelson (2006), reputations are temporary if the type of the player is determined once and for all at the beginning of the game. Reputations can have long-run implication by incorporating some mechanisms by which the uncertainty about types is continually replenished. Examples include Holmström (1982), Cole et al (1995), Mailath and Samuelson (2001), and Phelan (2001) where the type is governed by a stochastic process.

An alternative reputation mechanism under adverse selection is pioneered by Kreps and Wilson (1982), Milgrom and Roberts (1982) and, Kreps et al. (1982). The notion of a "commitment" type³, long-run players who always play the same action, is introduced. An important commitment type is the Stackelberg type. The Stackelberg type is a longrun players committing to the Stackelberg action. As long as the future returns from maintaining a reputation from playing the Stackelberg action exceeds short-run returns from playing other actions, the equilibrium strategy for an ordinary player will be to try to acquire a reputation by masquerading as a Stackelberg type . This issue is treated in detail in Mailath and Samuelson (2006).

Hörner (2002) presents a model where competition can generate a reputation mechanism under adverse selection through repeated interactions between firms and consumers.

³Also called "Irrational", "Action", "Behavioural" or "Crazy" types.

As in Holmström (1982), quality is a noisy signal of a firm's efforts in this model. However, Hörner (2002) shows that competition generates an outside option for consumers to leave a firm when they believe that the firm belongs to a bad type. This poses a threat to a firm and induces it to constantly exert high efforts.

There has been a growing body of literature focusing on the economics of brands and reputations. Brands are viewed as carriers of reputations and can be traded. Tadelis (1999) explores this idea and formulates a basic framework to analyse the market of reputation. His paper shows that this market is indeed sustainable. In addition, it provides all agents both "old" and "young" with equal incentives to exert effort (Tadelis 2002). However, Mailath and Samuelson (2001) show that the long-run effects of such a market are complicated as good reputations are likely to be purchased and depleted by "inept" agents. Kreps (1990) presents an argument for trading reputations in a moral hazard type of model.

Cabral (2000) adopts the framework from Tadelis (1999) to study the brand stretching decision of a firm when it launches a new product. The paper shows that brand stretching signals high quality; better firms tend to launch new products under the same brand.

The theory of optimal stopping has been applied in many areas of economics. For example, Dutta and Rustichini (1993) study optimal stopping games, a variant of stochastic games, where each player has an irreversible action. The paper shows the existence of stopping equilibria and presents two applications of product innovations and asset sales. Feng and Gallego (1995) study the optimal time for a single price change from a given initial price. The paper derives conditions under which the firm should decrease or increase the price. This can be considered as time optimal time to start an "end-of-season" sale or stop promotional pricing.

However, the problem studied in this paper is slightly different from the standard single time stopping problem. The model presented in the paper allows the firm to stop the product and start a new one. Therefore, any optimal stopping (or starting) rule⁴ must account for this. In operation research literature, this type of problem is called a restart problem according to Katehakis and Vienott (1987). It has been shown by Gittins and Jones (1974), Whittle (1980,1982) and Variaya et al. (1985) that this sequence of a one-dimensional stopping problem (where only one project or product is being considered in each period) is much simplified but equivalent to the multi-armed bandit problem (the problem where at each stage there are many choices of possible actions or experiments,

⁴In this thesis, the words discontinuation and stopping or policy and rule are used interchangeably.

and a choice of action j results in an observation being taken from a jth experiment; the goal is to maximise the present value of an infinite stream of rewards).

3 The Basic Quality Choice Model with Consumer Learning

The basic model presented in this section is adopted and simplified from Holmström (1982)'s career concerns model. The difference is that, in Holmström (1982), an agent (a firm) chooses an action (quality choice) in every period but in this model product's quality is decided by a firm only once at the initial period and remains unchanged until the product is discontinued. Another major extension, which will be considered in later sections is the incorporation of options for a firm to discontinue a product and relaunch a new one. Therefore, the model in this section provides the basis for this extension.

Consider a simple discrete time model where a firm offers a single product for sale in a market consisting of a large number of identical consumers. At the beginning of the first period, a firm chooses once and for all its desired level of product quality, q. However, the actual product quality is $\tilde{q} = q + \gamma$ where γ is an idiosyncratic quality disturbance term and can be viewed as a random term that determines how the product matches the consumer's taste. The firm wants to produce a good quality product and satisfy consumer needs, but some characteristics of the product may not fit well with consumer's taste or may not be what the consumer actually wants. Therefore, the actual product quality is a combination of the true or intrinsic quality that the firm produces and a match with the consumer's taste. Example of this type of product includes brand name clothes where fashion designers produce good quality clothes with styles that may or may not match the consumer's taste. Other examples includes product with sophisticated technology and features such as cars, mobile phones or laptop computers. These products come with many features and vary in quality. Consumers will learn whether every feature suits with their preference when the products is purchased and used. It is assumed that γ is normally distributed with a zero mean and variance σ_q^2 .

Consumers are identical and live for only one period. In each period, new consumers come to the market and decide whether to buy a firm's product and how much they are willing to pay for it. The quality perceived by consumers is a noisy signal of the actual quality. If consumers decide to buy the product from the firm, the utility gained by them is $u_t = \tilde{q} + \varepsilon_{u_t}$ where $\varepsilon_{u_t} \sim N(0, \sigma_u^2)$. The term ε_{u_t} can be interpreted as the consumers' random experience of using the product. It reflects the fact that consumers with identical taste using identical products can gain different utility levels depending on some random factors such as how the product is used and taken care of, knowledge of the product's specifications, and familiarity with the product's features and functions. Having realised the utility gained, consumers make an imperfect report of their experience, $r_t = u_t + \varepsilon_{r_t}$ where $\varepsilon_{r_t} \sim N(0, \sigma_r^2)$. The report is imperfect because consumers cannot precisely communicate their feeling about the product to the next generation of consumers. The sequence of reports, $\{r_s\}_1^t$, and, therefore, the firm's average quality report or *reputation*, $R_t = \frac{\sum_{s=1}^t r_s}{t}$, are publicly observable and, since consumers cannot directly observe the actual quality of the product, \tilde{q} , they use this information to decide how much they are willing to pay for the product.

In any period t, after observing R_{t-1} , consumers are willing to pay up to their expected utility from the product. That is,

$$p_{t} = E [u_{t} | R_{t-1}]$$

= $E [q | R_{t-1}] + E [\gamma | R_{t-1}].$ (3.1)

The firm is risk neutral choosing quality level q^* to maximise the expected sum of profit over product life T.

$$q^* = \arg\max_{q} \sum_{t=1}^{T} \delta^{t-1} \left[E\left[p_t \right] - C\left(q \right) \right]$$
(3.2)

where $\delta \in (0, 1)$ is a discount factor⁵, C(q) is the unit cost of producing a product with quality q, and C(q) is increasing and convex in q. Note that, even though the market is not able to observe the firm's choice of quality, q, directly, it can be inferred by solving (3.2). Therefore, at period t + 1, observing r_t in equilibrium will be equivalent to observing the sequence

$$z_t \equiv r_t - q^* = q + \gamma + \varepsilon_{u_t} + \varepsilon_{r_t} - q^*$$

= $q + \gamma + \varepsilon_t - q^*$ (3.3)

At equilibrium $q = q^*$, then

$$z_t = \gamma + \varepsilon_t \tag{3.4}$$

⁵It is assumed that δ cannot be equal to 1; to avoid complication that may arise in defining value function for an optimal product stopping problem, see Section 5.1.

where the composite noise, $\varepsilon_t \sim N(0, \sigma_{\varepsilon}^2 = \sigma_u^2 + \sigma_r^2)$, is the combination of two error terms, $\varepsilon_{u_t} + \varepsilon_{r_t}$.

Consumers and the firm share a prior belief about γ . The prior distribution is assumed normally distributed with mean m_1 and variance σ_1^2 . Define the precision of ε_t as $h_{\varepsilon} = 1/\sigma_{\varepsilon}^2$ and precision of belief at time t as $h_t = 1/\sigma_t^2$ where σ_t^2 denotes variance of prior belief about γ at time t. The posterior distribution of γ will still be normal distribution with mean, variance and precision given by

$$m_{t+1} = \frac{h_t m_t + h_\varepsilon z_t}{h_t + h_\varepsilon} = \frac{h_1 m_1 + h_\varepsilon \sum_{s=1}^t z_s}{h_1 + th_\varepsilon}$$
(3.5)

$$h_{t+1} = h_{\varepsilon} + h_t = th_{\varepsilon} + h_1 \tag{3.6}$$

Consumers' willingness to pay (3.1) can be rewritten as

$$p_t = E[u_t | R_{t-1}] = q^* + E[\gamma | R_{t-1}] = q^* + m_t$$
(3.7)

The firm's problem is then to attain the supremum of

$$\sup_{q} \sum_{t=1}^{T} \delta^{t-1} \left[E\left[p_{t} \right] - C\left(q \right) \right]$$

or equivalently,

$$\sup_{q} \sum_{t=1}^{T} \delta^{t-1} \left[q^* + E[m_t] - C(q) \right]$$
(3.8)

using (3.5) and (3.3), the above expression becomes

$$\sup_{q} \sum_{t=1}^{T} \delta^{t-1} \left[q^* + \frac{h_1 m_1}{h_t} + \frac{h_{\varepsilon}}{h_t} \sum_{s=1}^{t-1} \left(E(\gamma) + q - q^* \right) - C(q) \right]$$

using (3.6) to get,

$$\sup_{q} \sum_{t=1}^{T} \delta^{t-1} \left[q^* + \frac{h_1 m_1}{h_t} + \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1} \left(m_1 + q - q^* \right) - C\left(q\right) \right]$$
(3.9)

The solution to (3.9) is then given by the first order condition (FOC)

$$\sum_{t=1}^{T} \delta^{t-1} \left[\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_{1}} - C'(q^{*}) \right] = 0$$
(3.10)

or,

$$\frac{1-\delta}{1-\delta^T} \sum_{t=1}^T \delta^{t-1} \left[\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1} \right] = C'(q^*)$$
(3.11)

From (3.11), the optimal level of quality depends upon the discount factor, the structure of the cost function, the precision of the random noise (the sum of the perceived quality noise and the imperfect report) and the precision of the first period prior belief as well as the product lifespan. When the firm becomes more patient or δ increases, it places greater weight on future benefits and futures become more important; the incentive to bias consumers' learning increases, and the more quality is exerted.

As the precision of the first period prior belief, h_1 , increases, the left-hand side (LHS) of (3.11) decreases. Since the convex cost structure is assumed, the firm will choose a lower level of quality in equilibrium. The reason for this is that, since the true product's quality is unobservable, consumers have to form beliefs about the product. A higher level of perceived quality influences perceptions about the quality disturbance term. The firm has an incentive to jam the signal by choosing higher quality in which it would potentially bias the process of the inference of consumers. As the precision of prior belief increases, this implicit incentive decreases and the lower product quality is optimal.

Recall that the composite error term, ε_t , is the sum of two idiosyncratic terms: consumers' utility noise, ε_{u_t} , and noise on consumer report, ε_{r_t} . The higher the variances of these two noise terms, the higher the variance of the composite error term, and hence the lower precision, h_{ε} . h_{ε} can be viewed as the parameter determining the speed of consumers' inference about actual product quality or how noisy is the environment. One can see this from (3.5),

$$m_{t+1} = \frac{h_t m_t + h_\varepsilon z_t}{h_t + h_\varepsilon}$$

If the variance of the composite noise term is high, the level of precision, h_{ε} , is low. The posterior belief of the actual quality is the weighted sum of prior and the new signal about quality. High h_{ε} implies that the new signal is not very informative and consumers should put less weight on this when updating their belief. In the extreme case, where the variance of the composite noise approaches infinity or $h_{\varepsilon} \to 0$, the new signal is totally uninformative and consumers cannot learn from the new information at all, $m_{t+1} \to m_t$. The LHS of (3.11) is higher if h_{ε} increases. The firm will produce a higher quality product. In a less noisy environment, the firm will have more incentive to produce a high quality product because the consumer will refer high-perceived utility and previous consumers' report to a high product quality. In contrast, under a noisy environment, consumers will believe that a high-perceived utility and report are the consequence of noisy terms and not the firm's actual product quality. Knowing this, the firm will produce a lower quality product. In this simple model, the only incentive for the firm to produce a high quality product is to mislead consumers' belief about the true quality of the product.

The quality offered by the firm increases with the length of time that the product is in the market. As one can easily verify, the LHS of (3.11) equals 0 when T is 1 and it approaches $(1 - \delta) \sum_{t=1}^{\infty} \delta^{t-1} [(t-1)h_{\varepsilon}/h_t] > 0$ as $T \to \infty$. When T is small, consumers have less information about product quality and the product is discontinued early before the actual quality is precisely learnt; the firm thus misses a lot of revenue that it would get if the selling period had increased. This lowers the incentive for the firm to produce a higher quality product and to signal jam the learning process of consumers. The firm has a higher incentive to produce a better quality product if the relationship between product and consumers lasts for a longer period because high quality can mislead consumers into thinking that the product is a better match for their taste (the product has high γ) and will increase their willingness to pay.

4 Decision to Discontinue

The model described in Section 3 shows how product quality is determined and assumes that the product will be discontinued after T periods have elapsed. This section extends the model to the case where a firm introduces a new product after a previous one has been discontinued. Two simple examples are presented. Section 4.1 considers the case where the product discontinuation period is deterministic, i.e. a product lifespan is fixed. In Section 4.2, the product lifespan is variable but the section is restricted to the case where the firm has only one chance to discontinue the product. The more general case, where the discontinuation period is variable and there is an unrestricted number of discontinuations will be analysed in Section 5.



Figure 1: Deterministic Product Discontinuation Policy

4.1 Deterministic Product Discontinuation Period

It has been shown that the true product quality is revealed almost completely as time progresses. It is interesting to ask, if a firm chooses to discontinue an old product and introduce a new one, what would be the optimal time for discontinuation and what quality level would the firm select? With the opportunity to launch a new product, the firm faces two decision problems, the optimal quality problem and the product discontinuation problem. This section considers the case where selling periods are determined in advance and are the same for every product. There are many industries where firms decide not to wait until the actual quality is fully realised. They release a new design for their products every a fixed length of periods and consumers are forced to start the process of learning over again. For example, some Japanese and German car manufacturers have policies to release a new car design every 5 or 7 years. Fashion designers release their new designs every 3 months when the season changes. This section assumes that the firm discontinues its old product and launches its new one every T period; Figure 1 illustrates this. This deterministic product lifespan assumption will be relaxed in the subsequent sections where variable discontinuation periods are allowed.

Whenever a firm releases its new product, the quality choices must satisfy the FOC (3.11):

$$\frac{1-\delta}{1-\delta^T}\sum_{t=1}^T \delta^{t-1} \left[\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right] = C'(q^*)$$

Since the selling period (T), the discount factor (δ) and the precision of the composite noise (h_{ε}) are the same for every product, the quality choice q^* for each product depends upon the precision of initial prior belief (h_1) . As the idiosyncratic quality terms, γ , for all products produced by the firm are normally distributed with zero mean and identical precision, it is reasonable for the consumer to have initial prior beliefs with the same mean and precision every time a firm launches a new product. The following assumption states this formally.

Assumption 4.1 Mean and precision of initial prior belief about quality noise at the time of launching are the same for every product i.e.,

$$m_1 = m_{t_0}^b$$
$$h_1 = h_{t_0}^b$$

for all new product b launched at time t_0 .

The following remark follows immediately from Assumption 4.1

Remark 4.2 Under Assumption 4.1, optimal quality choices are identical for all products.

In contract theory, γ can be viewed as a firm's type chosen randomly from a known distribution by nature. The quality choice chosen by a firm (an agent) can be viewed as its effort to signal jam and mislead consumer (or principal) learning of γ . In Holmström (1982), the manager's (agent's) ability or type is drawn by nature at the beginning and remains unchanged for all periods. In the limit, his ability will be fully known and the incentive for signal jamming will eventually disappear. The manager will exert less effort as his ability becomes more apparent. Holmström (1982)'s conclusion is that there exists no equilibrium in which a single firm repeatedly exerts high efforts. In contrast, the incentive to exert an effort in this model does not reduce as time progresses. If a firm chooses to launch a new product, quality noise for the new product will be drawn again from the same distribution function but is independent of the previous product's quality noise. Consumers have to start the whole process of learning again and the incentive for signal jamming for the firm and hence quality choice remains the same. In this model, a firm's type can change randomly by launching a new product. This high effort equilibrium is in line with what Mailath and Samuelson (1998) have emphasized; type changing is a device to obtain a sustained reputation effect. In fact, Holmström (1982) uses the same device by letting a firm's type follow a random walk process to obtain high effort equilibrium.

Let $y_t^a \equiv q_a^* + m_t^a - C(q_a^*)$ be the period t profit accrued to a firm from product a when the quality is chosen optimally at q_a^* . The total discounted return is

$$V = \sum_{i=1}^{T} \delta^{i-1} y_i^a + \delta^T \sum_{j=1}^{T} \delta^{j-1} y_j^b + \delta^{2T} \sum_{k=1}^{T} \delta^{k-1} y_k^c + \dots$$

The expected return is

$$E[V] = E\sum_{i=1}^{T} \delta^{i-1} y_i^a + \delta^T E\left[\sum_{j=1}^{T} \delta^{j-1} y_j^b + \delta^T \sum_{k=1}^{T} \delta^{k-1} y_k^c + \dots\right]$$

Since quality choices are identical for all products by Remark 4.2,

$$E[V] = E \sum_{i=1}^{T} \delta^{i-1} y_i + \delta^T E[V]$$

= $\frac{\sum_{t=1}^{T} \delta^{t-1} E[y_t]}{1 - \delta^T}$
= $\frac{\sum_{t=1}^{T} \delta^{t-1} [q^*(T) + E(m_t) - C(q^*(T))]}{(1 - \delta) \sum_{s=1}^{T} \delta^{s-1}}$

In equilibrium, $E(m_t) = m_1 = 0$. Then,

$$E[V] = \frac{[q^*(T) - C(q^*(T))]\sum_{t=1}^{T} \delta^{t-1}}{(1 - \delta)\sum_{s=1}^{T} \delta^{s-1}}$$
$$= \frac{q^*(T) - C(q^*(T))}{(1 - \delta)}$$
(4.1)

The expected return (4.1) shows that the exogenously fixed product lifespan has no direct influence on the value function. However, it has an indirect influence through optimal quality choice. From the FOC for quality choice (3.11) and the discussion at the end of Section (3), the optimal quality increases with product lifespan. Proposition 4.3 shows that it is optimal for a firm not to discontinue its first product.

Proposition 4.3 The optimal deterministic discontinuation period is to set $T \to \infty$.

Proof. From (4.1), the firm's problem is to choose T that achieves the supremum of the

expected return E[V]:

$$V^{*} = \sup_{T} \frac{q^{*}(T) - C(q^{*}(T))}{(1 - \delta)}$$

The convex cost structure implies that the expected return is concave in q^* and maximised at $C'(q^*(T)) = 1$. From the FOC of quality choice (3.11),

$$\frac{1-\delta}{1-\delta^T}\sum_{t=1}^T \delta^{t-1}\left(\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right) = C'\left(q^*\left(T\right)\right)$$

the RHS, which is the marginal return from quality is increasing in T. Then,

$$\sup_{T} \frac{1-\delta}{1-\delta^{T}} \sum_{t=1}^{T} \delta^{t-1} \left(\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_{1}} \right) = (1-\delta) \sum_{t=1}^{\infty} \delta^{t-1} \left(\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_{1}} \right)$$

But since $\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1} \leq 1$, then

$$(1-\delta)\sum_{t=1}^{\infty}\delta^{t-1}\left(\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_{1}}\right) = \lim_{T\to\infty}C'\left(q^{*}\left(T\right)\right) \le 1$$

It has been shown that since marginal return from quality is increasing with T, when $T \to \infty$ the marginal return from quality is highest but still less than 1 and this gives a firm the highest possible expected return.

This model of consumer learning suggests that a product with a longer selling period is of better quality. The reason for this is that the firm has a higher incentive to produce high quality product in order to bias consumer learning about the product quality noise term and increase consumer willingness to pay. However, this model also suggests that the firm should not discontinue the product and start a new one. According to the model, it is better to produce one product with a relatively long life than to produce many shorter life products. It seems that this simple model cannot explain why different firms choose different selling periods for their products. Incorporating demand or competition structures would enrich the present analysis. This issue is left for future research.

Note that extensions to the model, such as a fixed cost for introducing the new product or adding the waiting periods before the new product can be introduced, do not change above analysed results. The quality choice FOC (3.11) stays the same while the expected returns change slightly to

$$E[V] = \frac{[q^*(T) - C(q^*(T))] \sum_{t=1}^{T} \delta^{t-1} - F}{(1-\delta) \sum_{s=1}^{T+S} \delta^{s-1}}$$

where F is a fixed cost for launching the new product and S is the length of periods the firm has to wait before the new product can be launched after stopping the old one. These extensions do not affect quality choice but lower the expected returns.

4.2 When to Start a New Product? A Simple Illustration

In the previous section, the product discontinuation period is known and fixed in advance and the firm has to stop its product; even belief about product quality noise is high. This seems unreasonable. In general, the firm would be willing to stop its product when the product is believed to have low quality and consumers are willing to pay less for it. Therefore, the discontinuation period varies depending on belief about product quality and profit prospect given available information in each period. This section presents a simple illustration of the problems where the firm has only *one chance* to stop an old product and start a new one to show how the optimal product stopping time is determined and how this decision could or could not affect the quality choice. The structure of a more complicated decision problem where the firm can discontinue an old product and start a new one at any period and as many times as it wants will be discussed in Section 4.3. The optimal product discontinuation policy for this problem, which is one of the central questions of this paper, will be analysed in details in Section 5.

Suppose that the selling period is not fixed in advance, the quality choice chosen by the firm must satisfy the following FOC

$$(1-\delta)\sum_{t=1}^{\infty}\delta^{t-1}\left(\frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_{1}}\right) = C'(q^{*})$$

$$(4.2)$$

If a firm has only one chance to discontinue its product a and start the new product b, it will stop the product a at any time t only when expected profit from a conditional on information available in period t is lower than the expected profit from launching b. At time $t = t_0$, if firm continues to produce product a with quality q_a^* , it would get the

expected profit of

$$\begin{aligned} \pi_{a} &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[E\left[q_{a}^{*} + m_{t} \left| t = t_{0} \right] - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + E\left[m_{t} \left| t = t_{0} \right] - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + E\left[\frac{h_{1}m_{1} + h_{\varepsilon} \sum_{s=1}^{t-1} z_{s}}{h_{t}} \right] - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + \frac{h_{1}m_{1}}{h_{t}} + \frac{h_{\varepsilon}}{h_{t}} \sum_{s=1}^{t-1} z_{s} + \frac{h_{\varepsilon}}{h_{t}} \sum_{s=t_{0}}^{t-1} E\left[z_{s} \left| t = t_{0} \right] - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + \frac{h_{t_{0}}}{h_{t}} \left[\frac{h_{1}m_{1}}{h_{t_{0}}} + \frac{h_{\varepsilon}}{h_{t_{0}}} \sum_{s=1}^{t-1} z_{s} \right] + \frac{h_{\varepsilon}}{h_{t}} \sum_{s=t_{0}}^{t-1} m_{t_{0}} - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + \frac{h_{t_{0}}m_{t_{0}}}{h_{t}} + \frac{\left(t - t_{0} \right)h_{\varepsilon}}{h_{t}} + h_{1}} m_{t_{0}} - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + \left[\frac{\left(t_{0} - 1 \right)h_{\varepsilon} + h_{1}}{(t-1)h_{\varepsilon} + h_{1}} + \frac{\left(t - t_{0} \right)h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_{1}} \right] m_{t_{0}} - C\left(q_{a}^{*} \right) \right] \\ &= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{a}^{*} + m_{t_{0}} - C\left(q_{a}^{*} \right) \right] \end{aligned}$$

If, however, at $t = t_0$, it launches a new product *b* with quality q_b^* , the optimal quality must satisfy the FOC (4.2). Suppose the initial period prior belief of the idiosyncratic quality disturbance of product *b*, γ^b , has mean $m_{t_0}^b$ and precision $h_{t_0}^b$. Optimal quality must satisfy the following FOC:

$$(1-\delta)\sum_{t=t_0}^{\infty} \delta^{t-t_0} \left[\frac{(t-t_0)h_{\varepsilon}}{h_t}\right] = C'(q_b^*)$$

$$(1-\delta)\sum_{t=t_0}^{\infty} \delta^{t-t_0} \left[\frac{(t-t_0)h_{\varepsilon}}{(t-t_0)h_{\varepsilon} + h_{t_0}^b}\right] = C'(q_b^*)$$

$$(4.4)$$

Comparing (4.4) with (4.2), one can see that q_b^* will be higher than q_a^* if the initial prior belief of idiosyncratic quality disturbance of *b* has higher precision. Or, $q_a^* \leq q_b^* \leftrightarrow h_1 \geq h_{t_0}^b$. However, if $h_1 = h_{t_0}^b$ as assumed in Assumption 4.1, quality choices are identical. As stated in Remark 4.2, the firm chooses the same level of quality when launching a new product because all parameters are the same every time it makes the decision.

The firm will launch a new product in the period when the prior belief falls below some critical level. The following proposition states this.

Proposition 4.4 There exists a unique m^* such that a new product is launched at period t_0 if and only if $m_{t_0} < m^*$ and if all new products have the same initial prior belief as stated in Assumption 4.1, $m^* = m_1$.

Proof. Profit from launching a new product with quality q_b^* at time t_0 is

$$\pi_{b} = \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[E\left(q_{b}^{*}+m_{t}^{b}\right) - C\left(q_{b}^{*}\right) \right]$$

$$= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{b}^{*}+\frac{h_{t_{0}}^{b}m_{t_{0}}^{b}}{h_{t}} + \frac{h_{\varepsilon}}{h_{t}} \sum_{s=t_{0}}^{t-1} \left(m_{t_{0}}^{b}+q_{b}^{*}-q_{b}^{*} \right) - C\left(q_{b}^{*}\right) \right]$$

$$= \sum_{t=t_{0}}^{\infty} \delta^{t-t_{0}} \left[q_{b}^{*}+\frac{h_{t_{0}}^{b}m_{t_{0}}^{b}}{(t-t_{0})h_{\varepsilon} + h_{t_{0}}^{b}} + \frac{(t-t_{0})h_{\varepsilon}}{(t-t_{0})h_{\varepsilon} + h_{t_{0}}^{b}} m_{t_{0}}^{b} - C\left(q_{b}^{*}\right) \right]$$

$$(4.5)$$

At time $t = t_0$, a firm will stop old product a and launch new product b if $\pi_a \leq \pi_b$. Or from (4.3) and (4.5), one can show that

$$\begin{split} \sum_{t=t_0}^{\infty} \delta^{t-t_0} \left[q_a^* + m_{t_0} - C\left(q_a^*\right) \right] \\ &\leq \sum_{t=t_0}^{\infty} \delta^{t-t_0} \left[q_b^* + \frac{h_{t_0}^b m_{t_0}^b}{\left(t-t_0\right) h_{\varepsilon} + h_{t_0}^b} + \frac{\left(t-t_0\right) h_{\varepsilon}}{\left(t-t_0\right) h_{\varepsilon} + h_{t_0}^b} m_{t_0}^b - C\left(q_b^*\right) \right]. \end{split}$$

By Remark 4.2, the above inequality can be simplified to

$$\sum_{t=t_0}^{\infty} \delta^{t-t_0} m_{t_0} \le \sum_{t=t_0}^{\infty} \delta^{t-t_0} \left[\frac{h_{t_0}^b m_{t_0}^b}{(t-t_0) h_{\varepsilon} + h_{t_0}^b} + \frac{(t-t_0) h_{\varepsilon}}{(t-t_0) h_{\varepsilon} + h_{t_0}^b} m_{t_0}^b \right]$$
(4.6)

The LHS term of (4.6) is linear and increasing in m_{t_0} while the RHS term is constant in m_{t_0} . Then, there exists a unique $m_{t_0}^*$ that satisfies (4.6) with equality making firm indifferent from continuing old product or launching the new one. This proves the first part of Proposition 4.4.



Figure 2: Time path of m_t and product stopping at t_0

Assumption 4.1 implies

$$\sum_{t=t_0}^{\infty} \delta^{t-t_0} m_{t_0} \le \sum_{t=t_0}^{\infty} \delta^{t-t_0} m_1.$$
(4.7)

If all new products launched by the firm have the same initial prior belief about the quality disturbance term, the critical level m^* that makes the firm indifferent from launching a new or continuing with an old product is m_1 . This proves the second part of the proposition.

This means that the firm will choose to stop the old product and produce the new one immediately when prior belief about quality noise of the present period falls below the initial prior belief when a product is newly introduced. Figure 2 illustrates one of the possible time paths of m_t .

From the figure, if the firm cannot stop selling the product, the prior mean of a quality noise will converge to the true quality noise almost surely. However, if the firm can choose to discontinue the old product, it will choose to stop in the period t_0 when m_{t_0} falls below the initial prior mean about the quality noise of a newly launched product, m_1 . It is not necessary that a good quality product will survive. A high quality product can be judged as a product with lower than standard quality and can be discontinued after a series of unlucky events during the initial phases of its introduction.

In this simple case, the quality choice is not affected by the discontinuation decision because the firm has only one chance to launch the new product. In a more enriched setting where the firm can choose to stop its product at any time, the firm has to take into account the probability that the product will be discontinued under the optimal discontinuation rule when choosing quality. In addition, as will be shown in Section 5.4, optimal quality choice determines the expected profit from launching the new product and affects the optimal critical level m^* for which a product will be discontinued.

4.3 Structure of the Decision Problems

The quality choice and product stopping rule obtained in Section 4.2 are actually based on the assumption that a firm has only one chance of stopping the old product a and launching the new product b. After b has been introduced to the market, the firm will produce b forever. The structure of the problem is changed significantly if in every period the firm can decide whether to discontinue the old product and start the new one. Note that Assumption 4.1 will still be applied here and Remark 4.2 is still valid in this setting because the future looks the same every time the firm chooses to start the new product. However, this implies that the optimal stopping policy would be identical for every product as well.

Remark 4.5 Under Assumption 4.1, optimal product discontinuation policies are identical for all products.

The firm has two levels of decision; it starts by choosing quality choice, and then enters the product discontinuation game. Firstly, it has to choose a level of quality to maximise its expected profit from the product and q is the decision variable. Since a product could be discontinued by product discontinuation policy, the profitability of a product depends upon the policy implemented in the product discontinuation game. Then, quality choice, q, must be chosen optimally conditional on optimal product discontinuation policy being implemented. Similarly, the optimal discontinuation policy must be determined given the optimal level of quality, $q = q^*$. Therefore, solutions to the two decision problems will be determined simultaneously.

5 Product Discontinuation Policies

This section will show that optimal product stopping policies telling a firm when to launch a new product and maximises an expected discounted profit exist. The analysis starts by explaining the structure of the problem in Section 5.1. Section 5.2 shows that the rather complicated discontinuation problem involving stopping and restarting products can be re-expressed or transformed to a simple or typical stopping problem which is easier to work with. Then, Section 5.3 shows the existence of optimal product stopping policy to this problem. In Section 5.4, the optimal product stopping policy is derived and some comparative static results (holding an optimal quality choice constant) are presented in Section 5.5.

5.1 Structure of the Product Discontinuation Problem

A firm in this model has to make two decisions, when to optimally stop an old product and start a new one, and what would be the optimal quality choice for the new product. Given that quality is chosen optimally, in each period after observing m_t , a firm can choose whether to continue with the present product or launch the new one. The strategy of the firm is given by

$$\psi: \mathbb{R} \longrightarrow \{0, 1\}.$$

In this decision rule, 0 indicates a stoppage and 1 indicates the decision to continue the old product. The optimal product stopping policy consists of the sequence of these functions

$$\mathbf{\Psi}=(\psi_1,\psi_2,...)$$
 .

that maximises the total expected discounted return. It is assumed that the new product can be launched as many times as required and that the same stopping policy is used for each product. For any product a and at any time t, having observed a sequence $\{m_1^a, m_2^a, ..., m_t^a\}$ and received $\sum_{s=1}^t \delta^{s-1} [q_a^* + m_s^a - C(q_a^*)]$, the firm may want to discontinue product a and start with product b over again. Since the same stopping policy is used in every product and period and a sequence $\{m_1^a, m_2^a, ..., m_t^a\}$ is independent for each product, the optimal product lifespan for each product $T_a, T_b, ...$ is i.i.d. random variables. As in Section 4.1, let

$$y_t^a \equiv q_a^* + m_t^a - C\left(q_a^*\right)$$

be the period t profit accrued to firm from product a. For notation simplicity, superscript a may be dropped if it is not required in the context. The total discounted return is

$$V = \sum_{i=1}^{T_a} \delta^{i-1} y_i^a + \delta^{T_a} \sum_{j=1}^{T_b} \delta^{j-1} y_j^b + \delta^{T_a+T_b} \sum_{k=1}^{T_c} \delta^{k-1} y_k^c + \dots$$

The expected return is

$$E[V] = E \sum_{i=1}^{T_a} \delta^{i-1} y_i^a + E \delta^{T_a} E\left[\sum_{j=1}^{T_b} \delta^{j-1} y_j^b + \delta^{T_b} \sum_{k=1}^{T_c} \delta^{k-1} y_k^c + ...\right]$$
(5.1)
$$= E \sum_{i=1}^{T_a} \delta^{i-1} y_i^a + E \delta^{T_a} E[V]$$

$$= \frac{E \sum_{t=1}^{T_a} \delta^{t-1} y_t}{1 - E \delta^T}$$

$$E[V] = \frac{E \sum_{t=1}^{T_a} \delta^{t-1} y_t}{(1 - \delta) E \sum_{t=1}^{T_a} \delta^{t-1}}$$
(5.2)

The optimal product stopping policy is the rule that maximises E[V]. Let \Im denote the class of product discontinuation rules,

$$\Im = \left\{ T : T \ge 1, 0 < \sum_{i=1}^{T} \delta^{i-1} < \infty \right\}$$
(5.3)

The objective is to find a discontinuation policy $T \in \mathfrak{T}$ to achieve the supremum in

$$V^* = \sup_{T \in \Im} \frac{E \sum_{t=1}^T \delta^{t-1} y_t}{(1-\delta) E \sum_{t=1}^T \delta^{t-1}}$$
(5.4)

Two conditions are required for V^* to be well defined. First, an optimal product stopping policy must exist. This issue will be addressed in Section 5.3. Second, in order to avoid the problem of dealing with 0/0 or ∞/∞ , the following condition is required.

$$0 < \sum_{i=1}^{T} \delta^{i-1} < \infty \tag{5.5}$$

The left inequality of (5.5) is satisfied because the firm is assumed to sell the new product



Figure 3: Structure of the stopping problem

in the first period and observe at least one observation of z_t . The left inequality is also satisfied since the discount factor $0 < \delta < 1$ by assumption in Section 3. The structure of this stopping problem for any arbitrary product is shown in Figure 3. From the figure, at the start of each period t, a prior mean of product quality noise is known and the firm has to make a decision to stop or continue the product conditional on this state variable. If the firm chooses to continue the same product, it will receive state profit y_t and the value of random variable z_t is realised. Knowing z_t , the prior mean of product quality noise is updated to m_{t+1} and the firm enters period t + 1. Alternatively, if the firm chooses to stop the present product and start a new one, the whole process of learning will start all over again. If optimal discontinuation policy is employed, the firm will receive the total expected profit of V^* from this decision.

5.2 Original and Transformed Problems

This section illustrates how one can transform the product discontinuation problem (5.4) to a simpler problem and use the transformed problem to find an optimal product discontinuation policy and the value function of the original one. Section 5.3 shows that optimal product discontinuation policies for transformed and hence original problems exist.

The purpose of this section is to show that the solution of the problem presented in

Section 5.1 can be obtained from the solution of the following problem

$$\overline{V} = \sup_{T \in \mathfrak{S}} E\left(\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right)$$
(5.6)

for some λ . The following lemma shows that the two problems are related.

Lemma 5.1 If for some λ , $\sup_{T \in \mathfrak{S}} E\left(\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right) = 0$, then $\sup_{T \in \mathfrak{S}} \frac{E\sum_{t=1}^{T} \delta^{t-1} y_t}{(1 - \delta) E\sum_{t=1}^{T} \delta^{t-1}} = \lambda$

Proof. If $\sup_{T \in \mathfrak{F}} E\left(\sum_{t=1}^{T} \delta^{t-1} [y_t - \lambda (1 - \delta)]\right) = 0$ then for every stopping rule T, $E\left(\sum_{t=1}^{T} \delta^{t-1} [y_t - \lambda (1 - \delta)]\right) \leq 0$. It follows that

$$E\left(\sum_{t=1}^{T} \delta^{t-1} y_t - \lambda \left(1-\delta\right) \sum_{t=1}^{T} \delta^{t-1}\right) \le 0$$
$$\frac{E\sum_{t=1}^{T} \delta^{t-1} y_t}{\left(1-\delta\right) E\sum_{t=1}^{T} \delta^{t-1}} \le \lambda$$

and that

$$\sup_{T \in \Im} \frac{E \sum_{t=1}^{T} \delta^{t-1} y_t}{(1-\delta) E \sum_{t=1}^{T} \delta^{t-1}} = \lambda$$

The next proposition claims that any solution for (5.6) is also optimal for the original problem.

Proposition 5.2 If $\sup_{T \in \mathfrak{F}} E\left(\sum_{t=1}^{T} \delta^{t-1} [y_t - \lambda (1 - \delta)]\right) = 0$ is attained at T^* , then T^* is optimal for maximising $\sup_{T \in \mathfrak{F}} \frac{E\sum_{t=1}^{T} \delta^{t-1} y_i}{(1-\delta)E\sum_{t=1}^{T} \delta^{t-1}}$.

Proof. Suppose not, from Lemma 5.1, $\frac{E\sum_{t=1}^{T} \delta^{t-1} y_i}{(1-\delta)E\sum_{t=1}^{T} \delta^{t-1}} < \lambda$ and then

$$E\left(\sum_{t=1}^{T^*} \delta^{t-1} y_t - \lambda \left(1-\delta\right) \sum_{t=1}^{T^*} \delta^{t-1}\right) < 0$$
$$E\left(\sum_{t=1}^{T^*} \delta^{t-1} \left[y_t - \lambda \left(1-\delta\right)\right]\right) < 0$$

but this contradicts the fact that T^* is optimal and achieves the supremum in $\sup_{T\in\mathfrak{S}} E\left(\sum_{t=1}^T \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right) = 0$

Proposition 5.2 allows the transformation of the original maximisation problem (5.4) to the ordinary stopping problem (5.6) with the return when choosing to stop at time T being $\sum_{i=1}^{T} \delta^{i-1} [y_i - \lambda (1 - \delta)]$. If the optimal stopping rule for a transformed problem exists, then it is also optimal for the product stopping problem. From now on, the problem (5.4) will be referred to as the *original problem* and (5.6) as the *transformed problem*. Therefore, V^* and \overline{V} are the value function of the original and transformed problems respectively.

5.3 Existence of Optimal Product Discontinuation Policies

The objective of this section is to prove that the optimal product discontinuation policy for the problem described in Section 5.1 exists so that V^* expressed in (5.4) can be attained. It is shown in Lemma 5.1 from the previous section that the original product discontinuation problem can be transformed to problem (5.6) with an appropriate value of λ . In addition, by Proposition 5.2, if the optimal discontinuation policy for the transformed problem exists, then so does the original problem. The task now is to show that optimal stopping rules for problem (5.6) exist. The following lemmas show that the transformed product stopping problem possesses two basic properties that are sufficient for an optimal stopping rule to exist.

Lemma 5.3
$$E\left[\sup_{T\in\mathfrak{F}}\left(\sum_{t=1}^{T}\delta^{t-1}\left[y_t-\lambda\left(1-\delta\right)\right]\right)\right]<\infty$$

Proof.

$$\sup_{T \in \Im} \left(\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right] \right) = \sum_{t=1}^{T} \delta^{t-1} \left[q^* + m_t - C \left(q^* \right) - \lambda \left(1 - \delta \right) \right]$$

where $m_t = \frac{h_1 m_1 + h_{\varepsilon} \sum_{s=1}^{t-1} z_s}{h_1 + (t-1)h_{\varepsilon}}$ and $m_t | m_1$ is normally distributed with a zero mean and variance $\sigma_{m_t}^2 = (t-1) \left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1} \right]^2 \frac{h_1 + h_{\varepsilon}}{h_1 h_{\varepsilon}}$. (see Lemma 5.13 on page 40 below) Therefore,

$$E\left[\sup_{T\in\mathfrak{S}}\left(\sum_{t=1}^{T}\delta^{t-1}\left[y_{t}-\lambda\left(1-\delta\right)\right]\right)\right] = \sum_{t=1}^{T}\delta^{t-1}\left[q^{*}+E\left(m_{t}\right)-C\left(q^{*}\right)-\lambda\left(1-\delta\right)\right]$$
$$= \sum_{t=1}^{T}\delta^{t-1}\left[q^{*}-C\left(q^{*}\right)-\lambda\left(1-\delta\right)\right]$$
$$= \left[q^{*}-C\left(q^{*}\right)-\lambda\left(1-\delta\right)\right]\frac{1-\delta^{T}}{1-\delta}$$
$$< \infty$$

Lemma 5.4 $\limsup_{T\to\infty} \left(\sum_{t=1}^T \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right] \right) \leq \sum_{t=1}^\infty \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right]$ almost surely.

Proof. Since $\lim_{T\to\infty} \left(\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right] \right) = \sum_{t=1}^{\infty} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right]$ and, for a real-valued sequence, $\lim_{T\to\infty} s_T = s$ implies $\limsup_{T\to\infty} s_T = \liminf_{T\to\infty} s_T = s$. It follows that $\limsup_{T\to\infty} \left(\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right] \right) = \sum_{t=1}^{\infty} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right]$ almost surely and the lemma is proved.

Lemma 5.3 makes it possible to interchange summation and expectation in subsequent arguments of this section. It can be shown that Lemma 5.3 and Lemma 5.4 are sufficient for the existence of optimal stopping rules. The method of Chow and Robbins (1963) and Ferguson (2000) using the notion of regular stopping rules is applied to show the existence of optimal stopping rules. The following definition defines the concept of regular stopping rules introduced by Snell (1952).

Definition 5.5 A discontinuation policy T is said to be regular, if for every s,

$$E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right] > \sum_{t=1}^{s} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]$$

almost surely on $\{T > s\}$.

From Definition 5.5, stopping rule T is regular if it tells a firm to continue at time s and gets a higher expected return than from stopping at that state. The following lemma shows that for any stopping rule there is a regular one that gives at least as good an expected payoff.

Lemma 5.6 Given any discontinuation policy T, there is a regular discontinuation policy T' such that

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \ge E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right].$$

Proof. Define

$$T' = \min\left\{s \ge 0: E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right] \le \sum_{t=1}^{s} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right\}$$
(5.7)

That is, T' tells the firm to continue to use T and stop when T tells it to continue, but stop is at least as good. It is clear that T' is a discontinuation policy and that $T' \leq T$. On $\{T' = s\}$, $E\left[\sum_{t=1}^{T} \delta^{t-1} [y_t - \lambda (1-\delta)] \middle| m_1, ..., m_s\right] \leq \sum_{t=1}^{s} \delta^{t-1} [y_t - \lambda (1-\delta)]$ almost surely for all s, while on $\{T' = \infty\}$, it is true that $\sum_{t=1}^{T} \delta^{t-1} [y_t - \lambda (1-\delta)] = \sum_{t=1}^{T'} \delta^{t-1} [y_t - \lambda (1-\delta)] = \sum_{t=1}^{\infty} \delta^{t-1} [y_t - \lambda (1-\delta)]$ almost surely. Hence, for any $1 \leq T' < \infty$,

$$E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] + E\left[\sum_{t=T'+1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$
(5.8)

Consider the last term of (5.8), $E\left[\sum_{t=T'+1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$, since $T' \leq T$, the analysis can be separated into 2 cases.

Case I If T' = T, the last term of (5.8) disappears and then,

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

Case II If T' < T, according to the discontinuation policy defined in (5.7),

$$E\left[\sum_{t=1}^{T} \delta^{t-1} [y_t - \lambda (1-\delta)] \middle| m_1, ..., m_{T'}\right] - \sum_{t=1}^{T'} \delta^{t-1} [y_t - \lambda (1-\delta)] \le 0$$

$$\sum_{t=1}^{T} E\left(\delta^{t-1} [y_t - \lambda (1-\delta)] \middle| m_1, ..., m_{T'}\right) - \sum_{t=1}^{T'} \delta^{t-1} [y_t - \lambda (1-\delta)] \le 0$$

$$\sum_{t=T'+1}^{T} E\left(\delta^{t-1} [y_t - \lambda (1-\delta)] \middle| m_1, ..., m_{T'}\right) \le 0$$

$$E\left[\sum_{t=T'+1}^{T} \delta^{t-1} [y_t - \lambda (1-\delta)]\right] \le 0$$

and from (5.8), it follows that

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \ge E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

The interchange between summation and expectation is valid by Lemma 5.3. It remains to prove that T' is regular. Note that on $\{T' > s\}$,

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right] \ge E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right]$$

almost surely. Since

$$E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right] > \sum_{t=1}^{s} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]$$

almost surely on $\{T' > s\}$, then

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right] > \sum_{t=1}^s \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]$$

almost surely on $\{T' > s\}$.

The next lemma shows that, among regular optimal product discontinuation policies, the one that tells the firm to stop later yields no worse return. **Lemma 5.7** If T and T' are regular product discontinuation policies, then so is $T'' = \max\{T, T'\}$ and then

$$E\left[\sum_{t=1}^{T''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \ge \max\left\{E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right], E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]\right\}$$

Proof. If T' = T, $T'' = \max{\{T', T\}} = T' = T$, then

$$E\left[\sum_{t=1}^{T''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$
$$= E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

If $T' \neq T$, assuming without loss of generality that T' > T, so $T'' = \max \{T', T\} = T'$. Since T' is regular and T' > T,

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_T\right] > \sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]$$

almost surely. By the law of iterated expectation,

$$E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] > E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

and since T'' = T',

$$E\left[\sum_{t=1}^{T''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$
$$> E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

It remains to show that T'' is regular. Assuming again without loss of generality that T' > T so that T'' = T' and from above result

$$E\left[\sum_{t=1}^{T''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \ge E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

From regularity of T', it can be shown that

$$E\left[\sum_{t=1}^{T''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right] \ge E\left[\sum_{t=1}^{T'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right] \middle| m_1, ..., m_s\right]$$
$$> \sum_{t=1}^s \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]$$

almost surely on $\{T'' = T' > s\}$. This implies that T'' is regular.

By Lemma 5.6, the attention can be restricted just on regular product discontinuation policies. In addition, Lemma 5.7 shows that the set of all regular discontinuation policies is an ordered set. In other words, Lemma 5.7 enables one to rank between any pair of regular discontinuation policies. With these two lemmas, it is unable to show the existence of optimal product discontinuation policies for the firm. The following proposition, which shows the existence of optimal product discontinuation policy, is based on Ferguson (2000). The proposition is the application of Fatou Lemma which states that suppose that the sequence of functions is a sequence of random variables $X_{1,}X_{2,...}$ with $X_n \leq Y$ almost surely for some Y such that $E(|Y|) < \infty$, then $\limsup_{n\to\infty} EX_n \leq E\limsup_{n\to\infty} X_n$.

Proposition 5.8 There exists a product discontinuation policy T^* such that

$$E\left[\sum_{t=1}^{T^*} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = \overline{V}$$

where $\overline{V} = \sup_{T \in \Im} E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right].$

Proof. Let $\{T_1, T_2, \ldots\}$ be any sequence of stopping rule such that

$$\lim_{j \to \infty} E\left[\sum_{t=1}^{T_j} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = \overline{V}$$

From Lemma 5.6, There is always a sequence of regular rules $\{T'_1, T'_2, \ldots\}$ such that $\lim_{j\to\infty} E\left[\sum_{t=1}^{T'_j} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = \overline{V}$. Let $T''_j = \max\left\{T'_1, T'_2, \ldots, T'_j\right\}$, then by Lemma 5.7

$$E\left[\sum_{t=1}^{T_j''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \ge E\left[\sum_{t=1}^{T_j'} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

which implies that $\lim_{j\to\infty} E\left[\sum_{t=1}^{T_j''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = \overline{V}$. It follows that

$$\overline{V} = \lim \sup_{j \to \infty} E\left[\sum_{t=1}^{T_j''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$
(5.9)

and define $T^* = \sup \{T'_1, T'_2, \ldots\}$. Note that T''_j is a nondecreasing sequence of stopping policies converging to stopping rule T^* . Therefore, from Lemma 5.4, $\limsup_{j\to\infty} \sum_{t=1}^{T''_j} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right] \leq \sum_{t=1}^{T^*} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta \right) \right]$ almost surely. Taking expectation to both sides of the inequality yields

$$E\left[\lim\sup_{j\to\infty}\sum_{t=1}^{T_j''}\delta^{t-1}\left[y_t - \lambda\left(1-\delta\right)\right]\right] \le E\left[\sum_{t=1}^{T^*}\delta^{t-1}\left[y_t - \lambda\left(1-\delta\right)\right]\right]$$
(5.10)

Using Fatou Lemma, one can relate (5.9) and (5.10),

$$\overline{V} = \limsup_{j \to \infty} E\left[\sum_{t=1}^{T_j''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \le E\left[\limsup_{j \to \infty} \sum_{t=1}^{T_j''} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$
$$\le E\left[\sum_{t=1}^{T^*} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right]$$

But since $E\left[\sum_{t=1}^{T^*} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] \leq \overline{V}$ by the definition of \overline{V} ,

$$E\left[\sum_{t=1}^{T^*} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right] = \overline{V}$$

and the proposition is proved.

5.4 Optimal Product Discontinuation Policy

It was presented in the previous section that optimal discontinuation policies exist. The objective of this section is to derive the optimal product stopping policy that maximises the firm's expected total profit (5.1). Note that the structure of the product stopping problem described in Section 5.1 differs from ordinary stopping problems in two aspects. First, a firm can always launch a new product immediately after it chooses to stop the old product. Therefore, the new stopping problem is restarted again after the new product is launched and optimal stopping rules must consider this. This is different from ordinary stopping problems where the process ends after the decision maker chooses to stop. However, it was shown in Lemma 5.1 and Proposition 5.2 that the original restart problem (5.4) can be transformed to an ordinary stopping problem (5.6) with an appropriate value λ . Second, at any time t after selling product and gaining period profit of $y_t = q^* + m_t - C(q^*)$, random variable z_t is realised and the posterior distribution of quality noise is updated. The posterior mean, m_{t+1} , and hence period t+1 profit, $y_{t+1} = q^* + m_{t+1} - C(q^*)$, are known to the firm at the beginning of period t + 1 before the decision to stop or continue is made. This contrasts to ordinary stopping problems where payoffs from continuing are realised only after the decision to continue is made.

The objective for the original restart problem is to choose an optimal stopping policy, T^* , such that it maximises the expected total profit (5.1) or equivalently (5.2) which is rewritten here,

$$E[V] = \frac{E \sum_{t=1}^{T} \delta^{t-1} y_t}{(1-\delta) E \sum_{t=1}^{T} \delta^{t-1}}$$

The numerator term of the RHS of (5.2) is the expected discounted profit from one product accrued to the firm by using stopping rule T. The term $E \sum_{i=1}^{T} \delta^{i-1}$ in the denominator can be viewed as an expected discounted time used for one product to obtain the expected profit in the numerator. Since a firm can always launch a new product, the problem is an infinite horizon problem and $(1 - \delta)$ is multiplied to express the denominator term in normalised or *per period* form. Therefore, E[V] can be viewed not only as the expected discounted total profit over an infinite horizon but also as the expected average profit of firm per effective unit of time or the expected *rate of return*. It is important to note that the objective is, in effect, to choose an optimal stopping rule that maximises average profit or expected rate of return and *not* the expected discounted profit from a product. As discussed above, Lemma 5.1 enables the transformation of original problem (5.4),

$$V^* = \sup_{T \in \Im} \frac{E \sum_{t=1}^{T} \delta^{t-1} y_t}{(1-\delta) E \sum_{t=1}^{T} \delta^{t-1}}$$

to an ordinary stopping problem of the form in (5.6), $\sup_{T \in \mathfrak{F}} E\left(\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda \left(1 - \delta\right)\right]\right)$, for some value of λ . This transformed problem can be rewritten as

$$\sup_{T \in \Im} \left[E \sum_{t=1}^{T} \delta^{t-1} y_t - \lambda \left(1 - \delta \right) E \sum_{t=1}^{T} \delta^{t-1} \right]$$

and λ in this transformed problem can be explained as a shadow value of an effective unit of time measured in the same unit as the payoffs, and is as if the firm is being charged λ for each effective unit of time. In addition, according to Lemma 5.1 and Proposition 5.2, if λ is chosen at $\lambda = \lambda^*$ such that $\sup_{T \in \mathfrak{F}} E\left(\sum_{t=1}^T \delta^{t-1} [y_t - \lambda^* (1 - \delta)]\right) = 0$, the optimal expected total profit for the firm, V^* , will also equal to λ^* . In this case, λ^* can be viewed as the maximised expected total return over an infinite horizon that the firm can get from launching the new product using the optimal stopping policy.

To derive an optimal product discontinuation policy for a firm, the method given by Lemma 5.1 and Proposition 5.2 will be utilised. The strategy is to solve the transformed problem for an optimal stopping rule and the associated value function and choose λ^* so that the value function of the transformed problem is equal to zero. Then, the value function of the original problem, V^* , is equal to λ^* .

The next proposition shows that the optimal stopping rule for the transformed problem is to stop at the first t for which $m_t \leq m^*(\lambda, q^*)$ for some λ .

Proposition 5.9 Optimal product discontinuation policy T^* for the transformed problem (5.6), $\sup_{T \in \mathfrak{F}} E\left(\sum_{i=1}^T \delta^{i-1} [y_i - \lambda (1 - \delta)]\right)$, is in the form

$$T^* = \min \{t > 1 : m_t \le m^*(\lambda, q^*)\}$$

where $m^{*}\left(\lambda,q^{*}\right) = \lambda\left(1-\delta\right) - \left[q^{*}-C\left(q^{*}\right)\right]$

Proof. Suppose at time t, $\sum_{i=1}^{t-1} \delta^{i-1} [y_i - \lambda (1 - \delta)] = Y^*$ and it is optimal to stop. By the principal of optimality, $\sum_{i=1}^{t-1} \delta^{i-1} [y_i - \lambda (1 - \delta)]$ is at least as large as any future return $E\left[\sum_{i=1}^{T} \delta^{i-1} [y_i - \lambda (1 - \delta)]\right]$ for all stopping rules T or

$$\sum_{i=1}^{t-1} \delta^{i-1} [y_i - \lambda (1-\delta)] \ge E \left[\sum_{i=1}^T \delta^{i-1} [y_i - \lambda (1-\delta)] \right]. \text{ Then,}$$

$$\sum_{i=1}^{t-1} \delta^{i-1} [y_i - \lambda (1-\delta)] \ge \sum_{i=1}^{t-1} \delta^{i-1} [y_i - \lambda (1-\delta)] + E \left[\sum_{i=t}^T \delta^{i-1} [y_i - \lambda (1-\delta)] \right]$$

$$0 \ge \sum_{i=t}^T \delta^{i-1} [q^* + E (m_i) - C (q^*) - \lambda (1-\delta)]$$

$$0 \ge [q^* + m_t - C (q^*) - \lambda (1-\delta)] \left(\delta^{t-1} + \dots + \delta^{T-1} \right)$$

But since $(\delta^{t-1} + \dots + \delta^{T-1}) > 0$, then

$$0 \ge [q^* + m_t - C(q^*) - \lambda(1 - \delta)]$$
$$\lambda(1 - \delta) - [q^* - C(q^*)] \ge m_t$$
$$m^*(\lambda, q^*) \ge m_t$$

Then, it is optimal to stop only when $m_t \leq m^*(\lambda, q^*)$ and the optimal rule T^* must be of the form $T^* = \min\{t > 1 : m_t \leq m^*(\lambda, q^*)\}.$

The following corollary states the optimal product discontinuation policy for the original problem.

Corollary 5.10 Under optimal product discontinuation policy, a firm continues a product in period t if profit from continuation in period t is at least as large as the average optimal profit, or otherwise discontinues.

Proof. Let λ^* be the chosen value of λ such that the value function of the transformed problem is maximised at 0, that is

$$\overline{V} = \sup_{T \in \mathfrak{S}} E\left[\sum_{i=1}^{T} \delta^{i-1} \left[y_i - \lambda^* \left(1 - \delta\right)\right]\right] = 0$$

From Proposition 5.9, the optimal stopping rule for the transformed problem is $T^* = \min\{t > 1 : m_t \le m^*(\lambda^*, q^*)\}$. However, from Proposition 5.2, T^* is also the optimal product discontinuation policy for the firm and this achieves the supremum in (5.4),

$$V^* = \sup_{T \in \mathfrak{F}} \frac{E \sum_{i=1}^T \delta^{i-1} y_i}{(1-\delta) E \sum_{i=1}^T \delta^{i-1}}$$

Since λ^* is chosen such that $\overline{V} = 0$, by Lemma 5.1, $V^* = \lambda^*$. Therefore, under the optimal product discontinuation policy, the firm will stop the old product and launch the new one at period t if

$$m_{t} \leq m^{*} (\lambda^{*}, q^{*})$$

$$m_{t} \leq \lambda^{*} (1 - \delta) - [q^{*} - C (q^{*})]$$

$$q^{*} + m_{t} - C (q^{*}) \leq \lambda^{*} (1 - \delta)$$

$$q^{*} + m_{t} - C (q^{*}) \leq (1 - \delta) V^{*}$$
(5.11)

The LHS term of (5.11) is the firm's profit from continuing the product in period t. Since V^* is the optimal expected discounted total profit over the infinite horizon under the optimal discontinuation policy, the term $(1 - \delta) V^*$ in the RHS of (5.11) is an average or per period return under the implementations of the optimal discontinuation policy.

In other words, Corollary 5.10 shows that the optimal product discontinuation policy is in the following form

$$T^* = \min \{t > 1 : y_t \le \lambda^* (1 - \delta)\}$$

= min { $t > 1 : m_t \le \lambda^* (1 - \delta) - [q^* - C(q^*)]$ }

In still other words, the optimal product discontinuation policy, T^* , suggests the firm stops the first t when $m_t \leq m^*(\lambda^*, q^*)$ or $m_t \leq \lambda^*(1 - \delta) - [q^* - C(q^*)]$ where λ^* is the optimal expected discounted profit over the infinite horizon. Since m_t depend upon the realisation of random variables z_t , m_t are random variables. The optimal product discontinuation period is, then, a random variable and the firm can conjecture the probability that its current product will be continued at any given period t under the implementation of the optimal discontinuation policy. In fact, for any given discontinuation policy $T = \min\{t > 1 : m_t \leq m^*(\lambda, q^*)\}$ and product quality choice q, the firm can calculate the probability that its newly launched product will survive in any period t. Definition 5.11 defines this probability.

Definition 5.11 For any product discontinuation policy $T = \min \{t > 1 : m_t \leq m^*(\lambda, q^*)\}$ and any quality choice q, let κ_t be the probability that the current product will be continued in period t under the implementation of T. Then,

$$\kappa_t(q,\lambda) = P\left[\bigcap_{j \le t} (m_j \ge m^*(\lambda, q^*))\right]$$

A product will be continued in period t only when it survives the discontinuation policy in every previous period and κ_t denotes the probability of this survival. κ_t can be expressed in terms of distribution functions of random variables γ and ε_s for s = 1, ..., t. To show this, κ_2 , κ_3 and κ_4 will be derived. The expression of κ_t for any t > 1 will be subsequently derived in Lemma 5.12.

At the beginning of period t = 2, the market has realised random variable z_1 from period 1 and updated the prior belief of γ , $m_2 = \frac{h_1 m_1 + h_{\varepsilon} z_1}{h_1 + h_{\varepsilon}}$. Therefore, the probability that the firm will continue the product in period 2 is

$$\kappa_2(q,\lambda) = P[m_2 \ge m^*(\lambda,q^*)] = P\left[\frac{h_1m_1 + h_{\varepsilon}z_1}{h_1 + h_{\varepsilon}} \ge m^*(\lambda,q^*)\right]$$
$$= P\left[z_1 \ge m^*(\lambda,q^*) \cdot \left(\frac{h_1 + h_{\varepsilon}}{h_{\varepsilon}}\right)\right]$$

Recalling from (3.3) that $z_t = \gamma + q + \varepsilon_t - q^*$ and defining $\widetilde{m}_2 = m^* (\lambda, q^*) \cdot \left(\frac{h_1 + h_{\varepsilon}}{h_{\varepsilon}}\right)$,

$$\kappa_2(q,\lambda) = P\left[\gamma + q - q^* + \varepsilon_1 \ge \widetilde{m}_2\right]$$

= $P\left[\gamma + \varepsilon_1 \ge \widetilde{m}_2 - (q - q^*)\right]$
= $P\left[\varepsilon_1 \ge \widetilde{m}_2 - (\gamma + (q - q^*))\right]$
= $\int_{-\infty}^{\infty} \int_{\widetilde{m}_2 - (\gamma + q - q^*)}^{\infty} f(\gamma, \varepsilon_1) d\varepsilon_1 d\gamma$

by independence and normality assumptions,

$$\kappa_{2}(q,\lambda) = \int_{-\infty}^{\infty} \int_{\widetilde{m}_{2}-(\gamma+q-q^{*})}^{\infty} f(\gamma) \cdot f(\varepsilon_{1}) d\varepsilon_{1} d\gamma$$
$$= \int_{-\infty}^{\infty} \int_{\widetilde{m}_{2}-(\gamma+q-q^{*})}^{\infty} \frac{1}{(2\pi)\sigma_{q}\sigma_{\varepsilon}} \exp\left\{-\left(\frac{\gamma^{2}}{2\sigma_{q}^{2}}\right) - \left(\frac{\varepsilon_{1}^{2}}{2\sigma_{\varepsilon}^{2}}\right)\right\} d\varepsilon_{1} d\gamma$$

where $f(\cdot)$ is the distribution function of interested variables, i.e.,

$$f(\gamma) = \frac{1}{(2\pi)^{\frac{1}{2}} \sigma_q} \exp\left\{-\frac{\gamma^2}{2\sigma_q^2}\right\}$$

and

$$f(\varepsilon_1) = rac{1}{(2\pi)^{rac{1}{2}} \sigma_{\varepsilon}} \exp\left\{-rac{\varepsilon_1^2}{2\sigma_e^2}
ight\}.$$

For t = 3, the probability of old product continuation is

$$\begin{aligned} \kappa_3\left(q,\lambda\right) &= P\left[m_3 \ge m^*\left(\lambda,q^*\right) \cap m_2 \ge m^*\left(\lambda,q^*\right)\right] \\ &= P\left[\frac{h_1m_1 + h_{\varepsilon}(z_1 + z_2)}{h_1 + 2h_{\varepsilon}} \ge m^*\left(\lambda,q^*\right) \cap z_1 \ge m^*\left(\lambda,q^*\right) \cdot \left(\frac{h_1 + h_{\varepsilon}}{h_{\varepsilon}}\right)\right] \\ &= P\left[z_1 + z_2 \ge m^*\left(\lambda,q^*\right) \cdot \left(\frac{h_1 + 2h_{\varepsilon}}{h_{\varepsilon}}\right) \cap z_1 \ge m^*\left(\lambda,q^*\right) \cdot \left(\frac{h_1 + h_{\varepsilon}}{h_{\varepsilon}}\right)\right] \\ &= P\left[z_1 + z_2 \ge \widetilde{m}_3 \cap z_1 \ge \widetilde{m}_2\right] \end{aligned}$$

where $\widetilde{m}_j = m^* (\lambda, q^*) \cdot \left(\frac{h_1 + (j-1)h_{\varepsilon}}{h_{\varepsilon}}\right)$. Since $z_t = \gamma + q + \varepsilon_t - q^*$,

$$\kappa_{3}(q,\lambda) = P\left[2\left(\gamma+q-q^{*}\right)+\varepsilon_{1}+\varepsilon_{2}\geq \widetilde{m}_{3}\cap\gamma+q-q^{*}+\varepsilon_{1}\geq \widetilde{m}_{2}\right]$$
$$= P\left[\varepsilon_{2}\geq \widetilde{m}_{3}-\left(2\left(\gamma+q-q^{*}\right)+\varepsilon_{1}\right)\cap\varepsilon_{1}\geq \widetilde{m}_{2}-\left(\gamma+q-q^{*}\right)\right]$$
$$= \int_{-\infty}^{\infty}\int_{\widetilde{m}_{2}-(\gamma+q-q^{*})}^{\infty}\int_{\widetilde{m}_{3}-2(\gamma+q-q^{*})-\varepsilon_{1}}^{\infty}f\left(\gamma,\varepsilon_{1},\varepsilon_{2}\right)d\varepsilon_{2}d\varepsilon_{1}d\gamma$$

by independence and normality assumptions,

$$\begin{split} f\left(\gamma,\varepsilon_{1},\varepsilon_{2}\right) &= f\left(\gamma\right).f\left(\varepsilon_{1}\right).f\left(\varepsilon_{2}\right) \\ &= \frac{1}{\left(2\pi\right)^{\frac{3}{2}}\sigma_{q}\sigma_{\varepsilon}^{2}}\exp\left\{-\left(\frac{\gamma^{2}}{2\sigma_{q}^{2}}\right) - \left(\frac{\varepsilon_{1}^{2} + \varepsilon_{2}^{2}}{2\sigma_{\varepsilon}^{2}}\right)\right\} \end{split}$$

where $f(\cdot)$ is the distribution function of interested variables, i.e.,

$$f(\gamma) = \frac{1}{(2\pi)^{\frac{1}{2}} \sigma_q} \exp\left\{-\frac{\gamma^2}{2\sigma_q^2}\right\}$$

and

$$f(\varepsilon_s) = \frac{1}{(2\pi)^{\frac{1}{2}} \sigma_{\varepsilon}} \exp\left\{-\frac{\varepsilon_s^2}{2\sigma_{\varepsilon}^2}\right\}.$$

for s = 1, 2.

For t = 4, the probability of old product continuation is

$$\kappa_4(q,\lambda) = P\left[m_4 \ge m^*(\lambda,q^*) \cap m_3 \ge m^*(\lambda,q^*) \cap m_2 \ge m^*(\lambda,q^*)\right]$$
$$= P\left[\frac{h_1m_1 + h_{\varepsilon}(z_1 + z_2 + z_3)}{h_1 + 3h_{\varepsilon}} \ge m^*(\lambda,q^*) \cap m_3 \ge m^*(\lambda,q^*) \cap m_2 \ge m^*(\lambda,q^*)\right]$$
$$= P\left[\begin{array}{c}z_1 + z_2 + z_3 \ge m^*(\lambda,q^*)\left(\frac{h_1 + 3h_{\varepsilon}}{h_{\varepsilon}}\right) \cap z_2 + z_1 \ge m^*(\lambda,q^*)\left(\frac{h_1 + 2h_{\varepsilon}}{h_{\varepsilon}}\right)\\\cap z_1 \ge m^*(\lambda,q^*)\left(\frac{h_1 + h_{\varepsilon}}{h_{\varepsilon}}\right)\end{array}\right]$$

Since, again, $\widetilde{m}_j = m^* (\lambda, q^*) \cdot \left(\frac{h_1 + (j-1)h_{\varepsilon}}{h_{\varepsilon}}\right)$ and $z_t = \gamma + q + \varepsilon_t - q^*$

$$\kappa_4(q,\lambda) = P\left[z_1 + z_2 + z_3 \ge \widetilde{m}_4 \cap z_2 + z_1 \ge \widetilde{m}_3 \cap z_1 \ge \widetilde{m}_2\right]$$
$$= P\left[\bigcap_{j=1}^4 \left(\sum_{s=1}^{j-1} z_s \ge \widetilde{m}_j\right)\right]$$
$$= P\left[\bigcap_{j=1}^4 \left(\sum_{s=1}^{j-1} (\gamma + q - q^* + \varepsilon_s) \ge \widetilde{m}_j\right)\right]$$
$$= P\left[\bigcap_{j=1}^4 \left((j-1)(\gamma + q - q^*) + \sum_{s=1}^{j-1} \varepsilon_s \ge \widetilde{m}_j\right)\right]$$

$$\kappa_4(q,\lambda) = P\left[\bigcap_{j=1}^4 \left(\varepsilon_{j-1} \ge \widetilde{m}_j - (j-1)\left(\gamma + q - q^*\right) - \sum_{s=1}^{j-2} \varepsilon_s\right)\right]$$
$$= \int_{-\infty}^\infty \int_{\widetilde{m}_2 - (\gamma + q - q^*)}^\infty \int_{\widetilde{m}_3 - 2(\gamma + q - q^*) - \varepsilon_1}^\infty \int_{\widetilde{m}_4 - 3(\gamma + q - q^*) - (\varepsilon_1 + \varepsilon_2)}^\infty f\left(\gamma, \varepsilon_1, \varepsilon_2, \varepsilon_3\right) d\varepsilon_3 d\varepsilon_2 d\varepsilon_1 d\gamma$$

and $f(\gamma, \varepsilon_1, \varepsilon_2, \varepsilon_3)$ is the joint probability distribution function and

$$f(\gamma, \varepsilon_1, \varepsilon_2, \varepsilon_3) = f(\gamma) f(\varepsilon_1) f(\varepsilon_2) f(\varepsilon_3)$$
$$= \frac{1}{(2\pi)^{\frac{4}{2}} \sigma_q \sigma_{\varepsilon}^3} \exp\left\{-\left(\frac{\gamma^2}{2\sigma_q^2}\right) - \left(\frac{\varepsilon_1^2 + \varepsilon_2^2 + \varepsilon_3^2}{2\sigma_{\varepsilon}^2}\right)\right\}$$

by independence and normality assumptions.

Lemma 5.12 shows the expression of κ_t in terms of γ and ε_s for any $t > s \ge 1$.

Lemma 5.12 For any product discontinuation policy $T = \min\{t > 1 : m_t \le m^*(\lambda, q^*)\}$ and any quality choice q, the probability that the current product will be continued in period t under the implementation of T is

$$\kappa_t(q,\lambda) = \int_{-\infty}^{\infty} \int_{\widetilde{m}_2 - (\gamma + q - q^*)}^{\infty} \cdots \int_{\widetilde{m}_t - (t-1)(\gamma + q - q^*) - \sum_{s=1}^{t-2} \varepsilon_s}^{\infty} f(\gamma) f(\varepsilon_1) \cdots f(\varepsilon_{t-1}) d\varepsilon_{t-1} \dots d\varepsilon_1 d\gamma$$
(5.12)

where $\widetilde{m}_j = m^* \left(\lambda, q^*\right) \cdot \left(\frac{h_1 + (j-1)h_{\varepsilon}}{h_{\varepsilon}}\right),$

$$f(\gamma) = \frac{1}{(2\pi)^{\frac{1}{2}} \sigma_q} \exp\left\{-\frac{\gamma^2}{2\sigma_q^2}\right\}$$
$$f(\varepsilon_s) = \frac{1}{(2\pi)^{\frac{1}{2}} \sigma_\varepsilon} \exp\left\{-\frac{\varepsilon_s^2}{2\sigma_\varepsilon^2}\right\}$$

for $s \in \{1, 2, ..., t - 1\}$ and

$$f(\gamma) f(\varepsilon_1) \cdots f(\varepsilon_{t-1}) = \frac{1}{(2\pi)^{\frac{t}{2}} \sigma_q \sigma_{\varepsilon}^{t-1}} \exp\left\{-\left(\frac{\gamma^2}{2\sigma_q^2}\right) - \left(\frac{\varepsilon_1^2 + \cdots + \varepsilon_{t-1}^2}{2\sigma_{\varepsilon}^2}\right)\right\}$$

Proof. From Definition 5.11,

$$\kappa_t(q,\lambda) = P\left[\bigcap_{j \le t} \left(m_j \ge m^*(\lambda, q^*)\right)\right]$$
$$= P\left[\bigcap_{j \le t} \left(\frac{h_1m_1 + h_{\varepsilon}\sum_{s=1}^{j-1} z_s}{h_1 + (j-1)h_{\varepsilon}} \ge m^*(\lambda, q^*)\right)\right]$$

since $m_1 = 0$ and $\widetilde{m}_j = m^* (\lambda, q^*) \cdot \left(\frac{h_1 + (j-1)h_{\varepsilon}}{h_{\varepsilon}}\right)$,

$$\kappa_t(q,\lambda) = P\left[\bigcap_{j \le t} \left(\sum_{s=1}^{j-1} z_s \ge m^* \left(\lambda, q^*\right) \cdot \left(\frac{h_1 + (j-1)h_{\varepsilon}}{h_{\varepsilon}}\right)\right)\right]$$
$$= P\left[\bigcap_{j \le t} \left(\sum_{s=1}^{j-1} z_s \ge \widetilde{m}_j\right)\right]$$

Recalling from (3.3) that $z_t = \gamma + q + \varepsilon_t - q^*$,

$$\kappa_t (q, \lambda) = P \left[\bigcap_{j \le t} \left(\sum_{s=1}^{j-1} (\gamma + q - q^* + \varepsilon_s) \ge \widetilde{m}_j \right) \right]$$

$$= P \left[\bigcap_{j \le t} \left((j-1) (\gamma + q - q^*) + \sum_{s=1}^{j-1} \varepsilon_s \ge \widetilde{m}_j \right) \right]$$

$$= P \left[\bigcap_{j \le t} \left(\sum_{s=1}^{j-1} \varepsilon_s \ge \widetilde{m}_j - (j-1) (\gamma + q - q^*) \right) \right]$$

$$= P \left[\bigcap_{j \le t} \left(\varepsilon_{j-1} \ge \widetilde{m}_j - (j-1) (\gamma + q - q^*) - \sum_{s=1}^{j-2} \varepsilon_s \right) \right]$$

$$= \int_{-\infty}^{\infty} \int_{\widetilde{m}_2 - (\gamma + q - q^*)}^{\infty} \cdots \int_{\widetilde{m}_t - (t-1)(\gamma + q - q^*) - \sum_{s=1}^{t-2} \varepsilon_s}^{t-2} f(\gamma, \varepsilon_1, \dots, \varepsilon_{t-1}) d\varepsilon_{t-1} \dots d\varepsilon_1 d\gamma$$

where $f(\gamma, \varepsilon_1, \ldots, \varepsilon_{t-1})$ is the joint probability distribution function and

$$f(\gamma, \varepsilon_1, \dots, \varepsilon_{t-1}) = f(\gamma) f(\varepsilon_1) \cdots f(\varepsilon_{t-1})$$
$$= \frac{1}{(2\pi)^{\frac{t}{2}} \sigma_q \sigma_{\varepsilon}^{t-1}} \exp\left\{-\left(\frac{\gamma^2}{2\sigma_q^2}\right) - \left(\frac{\varepsilon_1^2 + \dots + \varepsilon_{t-1}^2}{2\sigma_{\varepsilon}^2}\right)\right\}$$

by independence and normality assumptions.

Having found the optimal discontinuation policy, $T^* = \min \{t > 1 : m_t \le m^*(\lambda^*, q^*)\}$ = $\min \{t > 1 : m_t \le \lambda^* (1 - \delta) - [q^* - C(q^*)]\}$, the remaining problem is to find the value of λ^* . However, as Lemma 5.1 suggested, λ^* is the value of λ that makes the optimal value function for the transformed problem equal to zero or,

$$\overline{V} = \sup_{T \in \mathfrak{S}} E\left[\sum_{t=1}^{T} \delta^{t-1} \left[y_t - \lambda^* \left(1 - \delta\right)\right]\right] = 0$$

It is important to show some results that will facilitate subsequent analysis. Some of the following are basic and standard results from the theory of probability.

Lemma 5.13 Conditional on the information at the time of launching new product, for any t > 1, quality choice q, and any $a \in \mathbb{R}$,

$$m_{t}\left|m_{1},q\sim N\left(\mu_{t}\left(q\right),\sigma_{m_{t}}^{2}\right)\right.$$

and at equilibrium quality, $q = q^*$,

$$m_t \left| m_1 \sim N\left(0, \sigma_{m_t}^2 \right) \right|$$

where $\mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}(q-q^*)$ and $\sigma_{m_t}^2 = (t-1)\left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right]^2 \frac{h_1+h_{\varepsilon}}{h_1h_{\varepsilon}}$

Proof. Conditional on information at time t = 1, the distribution of random variables $z_s = \gamma + q + \varepsilon_s - q^*$ for any s is normal with mean

$$E[z_s | t = 1] = E[\gamma | t = 1] + q - q^* = m_1 + q - q^*$$

but since initial prior belief $m_1 = 0$, $E[z_s | t = 1] = q - q^*$. For the variance, since γ and

 ε_s are independent, variance of z_s is

$$\begin{aligned} var\left(z_{s}\left|t=1\right.\right) &= var\left(\gamma\left|t=1\right.\right) + var\left(\varepsilon_{s}\left|t=1\right.\right) \\ &= \frac{1}{h_{1}} + \frac{1}{h_{\varepsilon}} \\ &= \frac{h_{1} + h_{\varepsilon}}{h_{1}h_{\varepsilon}} \end{aligned}$$

That is, $z_s \sim N\left(q - q^*, \frac{h_1 + h_{\varepsilon}}{h_1 h_{\varepsilon}}\right)$. Since $m_t = \frac{h_1 m_1 + h_{\varepsilon} \sum_{s=1}^{t-1} z_s}{(t-1)h_{\varepsilon} + h_1}$, m_t is a linear combination of normal distributed random variables z_s . Therefore, conditional on information at time t = 1, for any t > 1 m_t is normally distributed with mean:

$$E[m_t | t = 1] = \frac{h_{\varepsilon} \sum_{s=1}^{t-1} E[z_s | t = 1]}{(t-1)h_{\varepsilon} + h_1}$$
$$= \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1} (q-q^*)$$

and variance:

$$var\left[m_{t} \mid t=1\right] = \left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_{1}}\right]^{2} var\left[\sum_{s=1}^{t-1} z_{s} \mid t=1\right]$$

since z_s are independent

$$var[m_t | t = 1] = \left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1}\right]^2 \sum_{s=1}^{t-1} var[z_s | t = 1]$$
$$= (t-1) \left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1}\right]^2 \frac{h_1 + h_{\varepsilon}}{h_1 h_{\varepsilon}}$$

Therefore, $m_t | m_1, q \sim N\left(\mu_t\left(q\right), \sigma_{m_t}^2\right)$ where $\mu_t\left(q\right) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\left(q-q^*\right),$ $\sigma_{m_t}^2 = (t-1)\left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right]^2 \frac{h_1+h_{\varepsilon}}{h_1h_{\varepsilon}}.$ At equilibrium quality choice, $q^* = q, \mu_t\left(q\right) = 0$ for all t and the distribution of m_t con-

ditional on the information at time of launching at equilibrium quality is $m_t \mid m_1 \sim N \left(0, \sigma_{m_t}^2\right)$

Lemma 5.14 Conditional on the information at the time of launching the new product,

for any t > 1, quality choice q, and any $a \in \mathbb{R}$

$$P[m_t > a] = 1 - \Phi\left(\frac{a - \mu_t(q)}{\sigma_{m_t}}\right)$$

and at equilibrium quality, $q = q^*$

$$P\left[m_t > a\right] = 1 - \Phi\left(\frac{a}{\sigma_{m_t}}\right)$$

where $\mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}(q-q^*), \ \sigma_{m_t}^2 = (t-1)\left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right]^2 \frac{h_1+h_{\varepsilon}}{h_1h_{\varepsilon}} \ and \ \Phi(\cdot) \ is \ the \ standard \ normal \ cdf.$

Proof. From Lemma 5.13, $m_t | m_1, q \sim N(\mu_t(q), \sigma_{m_t}^2)$. Let $m_t = \mu_t(q) + \sigma_{m_t} v$ where $v \sim N(0, 1)$, then

$$P[m_t > a] = P[\mu_t(q) + \sigma_{m_t}v > a]$$
$$= P\left[v > \frac{a - \mu_t(q)}{\sigma_{m_t}}\right]$$
$$= 1 - \Phi\left(\frac{a - \mu_t(q)}{\sigma_{m_t}}\right)$$

At equilibrium quality choice, $q^{*} = q$, $\mu_{t}(q) = 0$ for all t

$$P\left[m_t > a\right] = 1 - \Phi\left(\frac{a}{\sigma_{m_t}}\right)$$

where $\Phi(\cdot)$ is the standard normal cdf.

Lemma 5.15 (Inverse Mills Ratio) Conditional on the information at the time of launching the new product, for any t > 1, quality choice q, and any $a \in \mathbb{R}$

$$E\left[m_{t} \mid m_{t} \geq a\right] = \mu_{t}\left(q\right) + \sigma_{m_{t}} \frac{\phi\left(\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}\right)}{1 - \Phi\left(\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}\right)}$$

and at equilibrium quality, $q = q^*$

$$E\left[m_{t} \mid m_{t} \geq a\right] = \sigma_{m_{t}} \frac{\phi\left(\frac{a}{\sigma_{m_{t}}}\right)}{1 - \Phi\left(\frac{a}{\sigma_{m_{t}}}\right)}$$

where $\mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}(q-q^*)$, $\sigma_{m_t}^2 = (t-1)\left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right]^2 \frac{h_1+h_{\varepsilon}}{h_1h_{\varepsilon}}$, $\phi(\cdot)$ and $\Phi(\cdot)$ are the standard normal pdf. and cdf. respectively.

Proof. Let $m_t = \mu_t(q) + \sigma_{m_t} v$ where $v \sim N(0, 1)$ and $\mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}(q-q^*)$, then

$$E\left[m_{t}|m_{t} \geq a\right] = \int_{a}^{\infty} m_{t} \cdot f\left(m_{t}|m_{t} \geq a\right) dm_{t}$$

$$= \int_{\frac{a-\mu_{t}(q)}{\sigma_{m_{t}}}}^{\infty} \left(\mu_{t}\left(q\right) + \sigma_{m_{t}}v\right) \frac{1}{\sigma_{m_{t}}} \frac{\phi\left(v\right)}{1 - \Phi\left(\frac{a-\mu_{t}(q)}{\sigma_{m_{t}}}\right)} d\sigma_{m_{t}}v$$

$$= \frac{1}{1 - \Phi\left(\frac{a-\mu_{t}(q)}{\sigma_{m_{t}}}\right)} \left[\mu_{t}\left(q\right) \int_{\frac{a-\mu_{t}(q)}{\sigma_{m_{t}}}}^{\infty} \phi\left(v\right) dv + \sigma_{m_{t}} \int_{\frac{a-\mu_{t}(q)}{\sigma_{m_{t}}}}^{\infty} v\phi\left(v\right) dv\right]$$

Since, for standard normal distribution, $\phi'(v) = -v\phi(v)$,

$$E\left[m_{t}|m_{t} \geq a\right] = \frac{1}{1 - \Phi\left(\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}\right)} \left[\mu_{t}\left(q\right)\int_{\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}}^{\infty}\phi\left(v\right)dv - \sigma_{m_{t}}\int_{\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}}^{\infty}\phi'\left(v\right)dv\right]$$
$$= \frac{1}{1 - \Phi\left(\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}\right)} \left[\mu_{t}\left(q\right)\left(1 - \Phi\left(\frac{a - \mu_{t}\left(q\right)}{\sigma_{m_{t}}}\right)\right) + \sigma_{m_{t}}\phi\left(\frac{a - \mu_{t}\left(q\right)}{\sigma_{m_{t}}}\right)\right]$$
$$= \mu_{t}\left(q\right) + \sigma_{m_{t}}\frac{\phi\left(\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}\right)}{1 - \Phi\left(\frac{a - \mu_{t}(q)}{\sigma_{m_{t}}}\right)}$$

At equilibrium quality choice, $q^* = q$, $\mu_t(q^*) = 0$ for all t and

$$E\left[m_{t} \mid m_{t} \geq a\right] = \sigma_{m_{t}} \frac{\phi\left(\frac{a}{\sigma_{m_{t}}}\right)}{1 - \Phi\left(\frac{a}{\sigma_{m_{t}}}\right)}$$

Recalling from Proposition 5.9 that the optimal product discontinuation policy is in

the form $T^* = \min\{t > 1 : m_t \leq m^*(\lambda, q^*)\}$ where $m^*(\lambda, q^*) = \lambda(1 - \delta) - [q^* - C(q^*)]$ and it achieves the supremum of $\overline{V}(\lambda) = \sup_{T \in \mathfrak{S}} E\left[\sum_{t=1}^T \delta^{t-1} [y_t - \lambda(1 - \delta)]\right]$, the value function of the transformed problem (5.6). One can find the value of $\overline{V}(\lambda)$ by considering the special case where the firm implements T^* and T^* tells the firm to continue the product forever. That is the case when $m_t \geq m^*(\lambda, q^*)$ for every $t \in \mathbb{N}$. The expected profit from this case must be $\overline{V}(\lambda)$ because this is the case when the firm implements T^* forever and, therefore, acts optimally in every period. Proposition 5.16 derives $\overline{V}(\lambda)$ using this argument.

Proposition 5.16 Let $T^* = \min \{t > 1 : m_t \le m^*(\lambda, q^*)\}$ be the optimal product discontinuation policy. The value function of the transformed problem (5.6) can be expressed in the following form

$$\overline{V}(\lambda) = \sum_{t=1}^{\infty} \kappa_t(\lambda) \, \delta^{t-1} \left[E\left(m_t | m_t \ge m^*(\lambda, q^*) \right) - m^*(\lambda, q^*) \right]$$

or

$$\overline{V}(\lambda) = \sum_{t=1}^{\infty} \kappa_t(\lambda) \,\delta^{t-1} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*(\lambda, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda, q^*)}{\sigma_{m_t}}\right)} - C\left(q^*\right) - \lambda\left(1 - \delta\right) \right) \tag{5.13}$$

where $m^*(\lambda, q^*) = \lambda (1 - \delta) - [q^* - C(q^*)], \ \sigma_{m_t}^2 = (t - 1) \left[\frac{h_{\varepsilon}}{(t - 1)h_{\varepsilon} + h_1}\right]^2 \frac{h_1 + h_{\varepsilon}}{h_1 h_{\varepsilon}} \ and \ \kappa_t(\lambda) \ is \kappa_t(q, \lambda) \ from \ Lemma \ 5.12 \ evaluated \ at \ q = q^*.$

Proof. From the definition of value function

$$\overline{V}(\lambda) = \sup_{T \in \mathfrak{S}} E\left(\sum_{i=1}^{T} \delta^{i-1} \left[y_i - \lambda \left(1 - \delta\right)\right]\right)$$

Since T^* is the optimal product discontinuation policy,

$$\overline{V}(\lambda) = E\left[\sum_{i=1}^{T^*} \delta^{i-1} \left[y_i - \lambda \left(1 - \delta\right)\right] \middle| m_i \ge m^*(\lambda, q^*)\right]$$

Consider the case when $m_t \ge m^*(\lambda, q^*)$ for every $t \in \mathbb{N}$ and the policy T^* calls a firm to continue forever. The value function, $\overline{V}(\lambda)$, is equal to the expected discounted profit over

the infinite horizon,

$$\overline{V}(\lambda) = \sum_{t=1}^{\infty} \kappa_t(\lambda) \,\delta^{t-1} E\left[q^* + m_t - C\left(q^*\right) - \lambda\left(1 - \delta\right)\right| m_t \ge m^*\left(\lambda, q^*\right)\right]$$

where $\kappa_t(\lambda) = \int_{-\infty}^{\infty} \int_{\tilde{m}_2-\gamma}^{\infty} \cdots \int_{\tilde{m}_t-(t-1)\gamma-\sum_{s=1}^{t-2}\varepsilon_s}^{\infty} f(\gamma) f(\varepsilon_1) \cdots f(\varepsilon_{t-1}) d\varepsilon_{t-1} \dots d\varepsilon_1 d\gamma$ is the probability that a firm will continue the product in period t derived by Lemma 5.12 where quality choice q has been set optimally at q^* . Since all terms in the bracket of the above expression are constant except random variables m_t , it can be rewritten as

$$\overline{V}(\lambda) = \sum_{t=1}^{\infty} \kappa_t(\lambda) \,\delta^{t-1} \left[q^* + E\left(m_t | m_t \ge m^*\left(\lambda, q^*\right)\right) - C\left(q^*\right) - \lambda \left(1 - \delta\right)\right]$$

By Lemma 5.13, $m_t | m_1 \sim N(0, \sigma_{m_t}^2)$ where $\sigma_{m_t}^2 = (t-1) \left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right]^2 \frac{h_1+h_{\varepsilon}}{h_1h_{\varepsilon}}$, and Lemma 5.15, it follows that

$$\overline{V}(\lambda) = \sum_{t=1}^{\infty} \kappa_t(\lambda) \,\delta^{t-1} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*(\lambda, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda, q^*)}{\sigma_{m_t}}\right)} - C\left(q^*\right) - \lambda\left(1 - \delta\right) \right)$$

and this proves the proposition.

Corollary 5.17 concludes the optimal product discontinuation policy and the value function for the original product discontinuation problem (5.4).

Corollary 5.17 The optimal product discontinuation policy and the value function of the original problem are, respectively, in the following forms,

$$T^* = \min\{t > 1 : m_t \le m^*(\lambda^*, q^*)\} = \min\{t > 1 : m_t \le \lambda^*(1 - \delta) - [q^* - C(q^*)]\}$$

and

$$V^* = \sup_{T \in \Im} \frac{E \sum_{t=1}^T \delta^{t-1} y_t}{(1-\delta) E \sum_{t=1}^T \delta^{t-1}} = \frac{E \sum_{t=1}^{T^*} \delta^{t-1} y_t}{(1-\delta) E \sum_{t=1}^T \delta^{t-1}} = \lambda^*$$

where λ^* satisfies the following equation

$$\sum_{t=1}^{\infty} \kappa_t \left(\lambda^*\right) \delta^{t-1} \left[E\left(m_t | m_t \ge m^*\left(\lambda^*, q^*\right)\right) - m^*\left(\lambda^*, q^*\right) \right] = 0$$
(5.14)

or

$$\sum_{t=1}^{\infty} \kappa_t \left(\lambda^*\right) \delta^{t-1} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*\left(\lambda^*, q^*\right)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*\left(\lambda^*, q^*\right)}{\sigma_{m_t}}\right)} - C\left(q^*\right) - \lambda^*\left(1 - \delta\right) \right) = 0$$
(5.15)

Proof. The proof is a straightforward application of Lemma 5.1 and Proposition 5.2 and hence omitted.

5.5 Comparative Static Analysis

This section analyses how the firm would change its optimal discontinuation policy when exogenous variables such as the precision of initial belief (h_1) , the informativeness of the noisy signal (h_{ε}) and the discount factor (δ) vary. It is important to note that the equilibrium optimal discontinuation policy and the equilibrium quality, considered in Section 6, are determined simultaneously. The comparative static results for both endogenous variables should be analysed together as there is a feedback link between them. However, for simplicity, the quality choice will be treated as a constant and the first order effect of variables mentioned above to the product discontinuation decision will be considered.

The condition determining the optimal discontinuation policy is (5.14) shown in Corollary 5.17, which can be rewritten here as (the arguments of $m^*(\lambda^*, q^*)$ and $\kappa_t(\lambda^*)$ are dropped and the terms are rewritten m^* and κ_t in all the below expressions to make the notations compact)

$$\sum_{t=1}^{\infty} \kappa_t \delta^{t-1} \left[E\left(m_t | m_t \ge m^* \right) - m^* \right] = \sum_{t=1}^{\infty} \kappa_t \delta^{t-1} \left(\sigma_{m_t} \frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - m^* \right) = 0 \quad (5.16)$$

Equation (5.16) where the Implicit Function Theorem will be applied is central for the comparative static analysis. Some important expressions are rewritten here as

$$\sigma_{m_t}^2 = (t-1) \left(\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1}\right)^2 \frac{h_1 + h_{\varepsilon}}{h_1 h_{\varepsilon}}$$
$$\kappa_t = \int_{-\infty}^{\infty} \int_{\widetilde{m}_2 - \gamma}^{\infty} \cdots \int_{\widetilde{m}_t - (t-1)\gamma - \sum_{s=1}^{t-2} \varepsilon_s}^{\infty} f(\gamma) f(\varepsilon_1) \cdots f(\varepsilon_{t-1}) d\varepsilon_{t-1} \dots d\varepsilon_1 d\gamma$$

$$\widetilde{m}_t = m^* \cdot \left(\frac{h_1 + (t-1) h_{\varepsilon}}{h_{\varepsilon}} \right)$$

and

$$E(m_t | m_t \ge m^*) = \sigma_{m_t} \frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*}{\sigma_{m_t}}\right)}$$

Terms κ_t and $E(m_t | m_t \ge m^*) - m^*$ in (5.14) are the functions of the optimal stopping policy (m^*) , the precision of the quality disturbance term (h_1) , and the precision of composite noise (h_{ε}) . It is important to understand how κ_t and $E(m_t | m_t \ge m^*) - m^*$ change as m^* , h_1 and h_{ε} change since this will be useful for the subsequent comparative static analysis. For κ_t , an increase in m^* raises \tilde{m}_t for all t. A higher \tilde{m}_t increases all the lower limits of integration of variable ε_t in the κ_t expression. This means that $\frac{\partial \kappa_t}{\partial \tilde{m}} < 0$ for all t. Therefore, the probability that a product will survive the discontinuation policy decreases. The same analysis can be carried out for h_1 and h_{ε} by examining how they affect \tilde{m}_t . Remark 5.18 restates these results

Remark 5.18 The probability that the product will be continued in period t, κ_t , decreases as

1. m^* increases or $\frac{\partial \kappa_t}{\partial m^*} < 0$ for all $m^* \in \mathbb{R}$, 2. h_1 increases or $\frac{\partial \kappa_t}{\partial h_1} < 0$ for all $h_1 \in \mathbb{R}^+$ and 3. h_{ε} decreases or $\frac{\partial \kappa_t}{\partial h_{\varepsilon}} > 0$ for all $h_{\varepsilon} \in \mathbb{R}^+$.

Remark 5.18.1 is simple and intuitive; if the policy for which the product will be continued is rigid, then the chances that it will survive such rule after the realisation of m_t are slim. For Remark 5.18.2, a higher h_1 implies that the market puts more weight on initial prior belief m_1 when forming prior belief m_t for future periods. As a result, m_t tends to lie closer to $m_1 = 0$ and the product is less likely to survive the discontinuation policy $m^* > 0$. It is an opposite story when h_{ε} is high. As h_{ε} gets higher, the market places more weight in the realisation of the new information z_t relative to initial prior belief m_1 . If h_{ε} is high or the environment is less noisy, the market would attribute the high realised value of z_t to the high level of quality disturbance term, γ , and it is more likely that z_t will be high for the future period t as well. Therefore, if the product can survive for some periods it is more likely that for each period t, the prior belief m_t will exceed m^* and the product survives the discontinuation policy. For $E(m_t | m_t \ge m^*) - m^*$, this term is always greater than 0 and one can verify that because m_t has zero mean, $[E(m_t | m_t \ge m^*) - m^*] \to \infty$ as $m^* \to -\infty$ and $[E(m_t | m_t \ge m^*) - m^*] \to 0$ as $m^* \to \infty$. Since m_t , in the perspective of the firm (see Lemma 5.13), is normally distributed and one can plot $[E(m_t | m_t \ge m^*) - m^*]$ against m^* . Figure 4 illustrates this relationship and shows that $[E(m_t | m_t \ge m^*) - m^*]$ is decreasing in m^* . Note that Figure 4 actually plots $\frac{E(m_t | m_t \ge m^*) - m^*}{\sigma_{m_t}}$ against $\frac{m^*}{\sigma_{m_t}}$ but since

$$E(m_t | m_t \ge m^*) - m^* = \sigma_{m_t} \frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - m^*$$
$$= \sigma_{m_t} \left(\frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - \frac{m^*}{\sigma_{m_t}}\right)$$
$$= \sigma_{m_t} \left(\frac{E(m_t | m_t \ge m^*)}{\sigma_{m_t}} - \frac{m^*}{\sigma_{m_t}}\right)$$
(5.17)

and σ_{m_t} is not influenced by m^* , then the figure still represents the relationship between $[E(m_t | m_t \ge m^*) - m^*]$ and m^* when both axes are rescaled.

Remark 5.19 $[E(m_t | m_t \ge m^*) - m^*]$ is decreasing in m^* or $\frac{\partial [E(m_t | m_t \ge m^*) - m^*]}{\partial m^*} < 0$ for all $m^* \in \mathbb{R}$.

To analyse the effects of h_1 and h_{ε} on $E(m_t | m_t \ge m^*) - m^*$, equation (5.17) suggests that h_1 and h_{ε} influence $E(m_t | m_t \ge m^*) - m^*$ via σ_{m_t} and also $\left(\frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - \frac{m^*}{\sigma_{m_t}}\right)$. Lemma 5.20 shows how h_1 and h_{ε} influence σ_{m_t} .

Lemma 5.20 The standard deviation of m_t , σ_{m_t} , conditional on information at the time of product launching **decreases** if

- 1. h_1 increases
- 2. h_{ε} increases for $t > 3 + \frac{h_1}{h_{\varepsilon}}$ or h_{ε} decreases for $t < 3 + \frac{h_1}{h_{\varepsilon}}$.



Figure 4: Plot of $[E(m_t | m_t \ge m^*) - m^*]$

Proof. For the first part, the partial derivative of σ_{m_t} with respect to h_1 is

$$\begin{aligned} \frac{\partial \sigma_{m_t}}{\partial h_1} &= \frac{\partial}{\partial h_1} \left[(t-1) \left(\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon} + h_1} \right)^2 \frac{h_1 + h_{\varepsilon}}{h_1 h_{\varepsilon}} \right] \\ &= -\frac{1}{h_1^2} \frac{(t-1)h_{\varepsilon}}{((t-1)h_{\varepsilon} + h_1)^3} \left((t-1)h_{\varepsilon}^2 + 3h_{\varepsilon}h_1 + 2h_1^2 \right) \\ &< 0 \end{aligned}$$

For the second part, the partial derivative of σ_{m_t} with respect to h_{ε} is given by

$$\frac{\partial \sigma_{m_t}}{\partial h_{\varepsilon}} = (t-1) \frac{(3-t) h_{\varepsilon} + h_1}{\left((t-1) h_{\varepsilon} + h_1\right)^3}$$

This expression is less than zero if and only if $(3-t)h_{\varepsilon} + h_1 < 0$ or $t > 3 + \frac{h_1}{h_{\varepsilon}}$.

The reason for Lemma 5.20.1 is that, when the precision of the initial prior belief is high, the posterior belief m_t is close to prior and then the variation or standard deviation of m_t is small. Lemma 5.20.2 suggests that as h_{ε} increases it causes σ_{m_t} to increase in the first few periods depending on the values of h_1 and h_{ε} but it reduces σ_{m_t} for large t. If the signal is very informative (high h_{ε}) and after $t = 3 + \frac{h_1}{h_{\varepsilon}}$ periods, the market will be able to predict the quality disturbance term γ more precisely and m_t is more closed to γ and is less variable (low σ_{m_t}).

Since σ_{m_t} is influenced by h_1 and h_{ε} and according to the plot from Figure 4 that $\left(\frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1-\Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - \frac{m^*}{\sigma_{m_t}}\right)$ is decreasing in $\frac{m^*}{\sigma_{m_t}}$ (or increasing in σ_{m_t}), the relationship between $[E\left(m_t \mid m_t \ge m^*\right) - m^*] = \sigma_{m_t} \left(\frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1-\Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - \frac{m^*}{\sigma_{m_t}}\right)$ and h_1, h_{ε} can be obtained. Remark 5.21 summarises this.

Remark 5.21 $E(m_t | m_t \ge m^*) - m^*$ decreases if

- 1. h_1 increases and
- 2. h_{ε} increases for $t > 3 + \frac{h_1}{h_{\varepsilon}}$ or h_{ε} decreases for $t < 3 + \frac{h_1}{h_{\varepsilon}}$

When h_1 increases, for the reason explained above, the standard deviation of m_t for the future t decreases. This means that the expectation of m_t given that $m_t \ge m^*$ or $E(m_t|m_t \ge m^*)$ must increase. Figure 5 shows the probability distribution functions of m_t with a different standard deviation ($\sigma_{m_t} = 1$ and 2). The vertical axis is drawn at the point where $m_t = m^*$ (arbitrary set $m^* = 1$ in the figure). The figure shows that areas to the right of the vertical axis have different expected values ($E(m_t|m_t \ge m^*) = 1$. 5251 and 2.2822 respectively). When the precision of initial prior belief is higher, the future realisation of m_t tends to lie closer to m_1 and the firm cannot expect the high variation in m_t . Then, the expected return in period t, given that the product survives the discontinuation policy, will be lower.

Product discontinuation policy can be viewed as the policy for quality control. It is expressed in the form of minimal observed quality or reputation standard. Since the actual quality is unknown at the time of launching, the firm controls its product quality by monitoring consumers' experience from the product. This is reflected in the reputation of the product or the prior belief about the quality noise, m_t , in each period. A high m^* implies that the firm applies a rigid supervision on its product quality and it is more likely for the product to be discontinued. This idea is stated more formally in the next definition.



Figure 5: pdf functions of m_t with different σ_{m_t}

Definition 5.22 Given any product discontinuation policy $T = \min\{t > 1 : m_t \le m^T\}$, a product discontinuation policy $T' = \min\{t > 1 : m_t \le m^{T'}\}$ is said to be **a more stringent (lenient) quality control** if T' calls to discontinue the product sooner (later) almost surely or $m^{T'} > m^T$ ($m^{T'} < m^T$).

Comparative static results can be obtained using the Implicit Function Theorem. From (5.16), define

$$F(m^*, \delta, h_{1,h_{\varepsilon}}) = \sum_{t=1}^{\infty} \kappa_t \delta^{t-1} \left(\sigma_{m_t} \frac{\phi\left(\frac{m^*}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*}{\sigma_{m_t}}\right)} - m^* \right)$$

Proposition 5.23 shows that, when the firm becomes more patient, the product has a shorter life almost surely.

Proposition 5.23 As the discount factor (δ) increases, the optimal product discontinuation policy becomes more stringent.

Proof. The partial derivative of $F(m^*, \delta, h_1, h_{\varepsilon})$ with respect to the optimal discontinua-

tion policy m^* is

$$F_{m^*} = \sum_{t=1}^{\infty} \delta^{t-1} \left[\kappa_t \frac{\partial \left[E\left(m_t | m_t \ge m^* \right) - m^* \right]}{\partial m^*} + \left(E\left(m_t | m_t \ge m^* \right) - m^* \right) \frac{\partial \kappa_t}{\partial \widetilde{m}_t} \frac{\partial \widetilde{m}_t}{\partial m^*} \right] < 0$$

since $(E(m_t|m_t \ge m^*) - m^*) > 0$ for all $m^* \in \mathbb{R}$ and from Remark 5.18.1 and Remark 5.19, $\frac{\partial \kappa_t}{\partial \tilde{m}_t} \frac{\partial \tilde{m}_t}{\partial m^*} < 0$ and $\frac{\partial [E(m_t|m_t \ge m^*) - m^*]}{\partial m^*} < 0$. The partial derivative of $F(m^*, \delta, h_1, h_{\varepsilon})$ with respect to δ is

$$F_{\delta} = \sum_{t=1}^{\infty} (t-1) \, \delta^{t-2} \kappa_t \left[E(m_t | m_t \ge m^*) - m^* \right] \\> 0$$

By the Implicit Function Theorem, $\frac{dm^*}{d\delta} = -\frac{F_{\delta}}{F_{m^*}} > 0.$

The result shown in Proposition 5.23 seems counter-intuitive but this reveals an important implication: a patient firm will be impatient with its low quality products. When the firm is patient, it puts greater weight on futures, supplies better quality and gets a higher expected revenue and hence a higher expected per period revenue. The firm will stop the product in the period when the period revenue falls below the expected per period revenue, since the firm knows that it can get a higher revenue on average by launching the new product. Therefore, the product will be discontinued earlier almost surely.

Proposition 5.24 As the precision of initial prior belief (h_1) increases, the optimal product discontinuation policy becomes more lenient.

Proof. The partial derivative of $F(m^*, \delta, h_1, h_{\varepsilon})$ with respect to the precision of initial prior belief h_1 is

$$F_{h_1} = \sum_{t=1}^{\infty} \delta^{t-1} \left[\kappa_t \frac{\partial \left[E\left(m_t | m_t \ge m^* \right) - m^* \right]}{\partial h_1} + \left(E\left(m_t | m_t \ge m^* \right) - m^* \right) \frac{\partial \kappa_t}{\partial \widetilde{m}_t} \frac{\partial \widetilde{m}_t}{\partial h_1} \right]$$

From Lemma 5.20, $\frac{\partial [E(m_t|m_t \ge m^*) - m^*]}{\partial h_1} < 0$ and Remark 5.18, $\frac{\partial \kappa_t}{\partial \tilde{m}_t} \frac{\partial \tilde{m}_t}{\partial h_1} < 0$, it follows that $F_{h_1} < 0$. It was shown in the proof of Proposition 5.23 that $F_{m^*} < 0$. Therefore, by the Implicit Function Theorem, $\frac{dm^*}{dh_1} = -\frac{F_{h_1}}{F_{m^*}} < 0$

As the initial prior belief about the product quality noise becomes more precise, the expected return from the product decreases for two reasons. First, the probability that the product will survive the product discontinuation policy is lower because of the tendency that m_t will lie closer to m_1 . Second, the higher tendency that m_t will lie closer to m_1 from the product given that it survives the discontinuation policy. The smaller expected return from the product lowers the condition for which the product is discontinued.

The effect of h_{ε} on m^* is ambiguous. A higher h_{ε} clearly raises the probability that the product will be continued in every period. However, its effect on the expected profit, given that the product survives, is unclear as $E(m_t | m_t \ge m^*) - m^*$ increases for the first few periods but declines afterwards. The net effect relies on the parameter values of the model.

6 Quality Choices

The optimal product discontinuation policy and the value function derived in Corollary 5.17 depend upon the optimal level of quality choice, q^* . This section derives a condition for which the optimal quality is determined. Note that the optimal quality also depends on the optimal discontinuation policy and that both must be solved simultaneously using the condition shown in Corollary 5.17 and the one derived in this section.

A firm chooses a quality choice that maximises the product's lifetime expected profit on the basis that an optimal product discontinuation policy will be implemented after the product is launched. For any period t, the probability that a product with quality q will be continued under the implementation of an optimal discontinuation policy is $\kappa_t(q, \lambda^*)$. The expected profit from a product with quality q is given by

$$\sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(q, \lambda^* \right) \left[E\left(p_t \left| m_t \ge m^* \left(\lambda^*, q^* \right) \right) - C\left(q \right) \right] \right]$$

where p_t is consumers' willingness to pay defined in (3.7). The expectation is now conditional on events where the product survives the optimal discontinuation policy. Using (3.7), the above expression can be rewritten as

$$\sum_{t=1}^{\infty} \delta^{t-1} \kappa_t (q, \lambda^*) [q^* + E(m_t | m_t \ge m^*(\lambda^*, q^*)) - C(q)]$$

The firm's quality choice problem is

$$\sup_{q} \sum_{t=1}^{\infty} \delta^{t-1} \kappa_t (q, \lambda^*) [q^* + E(m_t | m_t \ge m^*(\lambda^*, q^*)) - C(q)]$$

From the first part of Lemma 5.15, the problem can be expressed as follows,

$$\sup_{q} \sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(q, \lambda^* \right) \left[q^* + \mu_t \left(q \right) + \sigma_{m_t} \frac{\phi \left(\frac{m^* \left(\lambda^*, q^* \right) - \mu_t \left(q \right)}{\sigma_{m_t}} \right)}{1 - \Phi \left(\frac{m^* \left(\lambda^*, q^* \right) - \mu_t \left(q \right)}{\sigma_{m_t}} \right)} - C \left(q \right) \right]$$
(6.1)

where $\mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}(q-q^*)$, $\sigma_{m_t}^2 = (t-1)\left[\frac{h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}\right]^2 \frac{h_1+h_{\varepsilon}}{h_1h_{\varepsilon}}$, $\phi(\cdot)$ and $\Phi(\cdot)$ are the standard normal pdf. and cdf. respectively.

The FOC for the quality problem (6.1) is presented in the following proposition

Proposition 6.1 For any $\lambda^* \in \mathbb{R}$ and any optimal product discontinuation policy $T^* = \min\{t > 1 : m_t \leq m^*(\lambda^*, q^*)\}$, the optimal product quality, q^* , satisfies the following FOC

$$\sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(\lambda^*\right) \left(\mu_t' - \mu_t' \frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} \left(\frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} - \frac{m^*\left(\lambda^*, q^*\right)}{\sigma_{m_t}}\right) - C'\left(q^*\right) \right) + \sum_{t=1}^{\infty} \delta^{t-1} \left. \frac{\partial}{\partial q} \kappa_t\left(q, \lambda^*\right) \right|_{q=q^*} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} - C\left(q^*\right) \right) = 0 \quad (6.2)$$

where $\kappa_t(\lambda^*) = \kappa_t(q,\lambda^*)$ is evaluated at $q = q^*$ and $\mu'_t = \frac{\partial}{\partial q}\mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}$.

Proof. According to product rule, the FOC for the problem (6.1) is

$$\sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(\lambda^*\right) \left(\mu_t' - \mu_t' \left(\frac{\phi'\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right) \left(1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)\right) + \phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right) \phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{\left(1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)\right)^2} \right) - C'\left(q^*\right) \right) + \sum_{t=1}^{\infty} \delta^{t-1} \left. \frac{\partial}{\partial q} \kappa_t\left(q, \lambda^*\right) \right|_{q=q^*} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} - C\left(q^*\right) \right) = 0$$

using the fact that for the standard normal distribution, $\phi'(v) = -v\phi(v)$, and re-scaled

yield

$$\sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(\lambda^*\right) \left(\mu_t' - \mu_t' \frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} \left(\frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} - \frac{m^*\left(\lambda^*, q^*\right)}{\sigma_{m_t}}\right) - C'\left(q^*\right) \right) + \sum_{t=1}^{\infty} \delta^{t-1} \left. \frac{\partial}{\partial q} \kappa_t\left(q, \lambda^*\right) \right|_{q=q^*} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} - C\left(q^*\right) \right) = 0$$

The Proposition 6.1 suggests that there are two channels by which quality choice can affect the expected return. The first term of the FOC (6.2) is the standard career concerns mechanism. The firm has an incentive to increase its product quality in order to mislead consumers that the product has a high unobservable quality because this will increase consumers' willingness to pay and the firm's profits. The second term of the FOC (6.2) represents the second incentive for the firm to increase quality: higher quality biases consumer inference about unobservable quality noise, and increases the probability that the product will be continued, which in turn increases expected revenue. This additional incentive to increase quality is the extension of the prototype career concerns model and occurs in the context where the relationship between principal and agent can be terminated by the realisation of the random variable. In both mechanisms the firm or agent exerts a greater effort to mislead consumer or principal inference and this bias has two effects. Firstly, it encourages principal to believe that the agent is of the better type and this will results in higher returns to the agent. Secondly, when the probability of relationship termination depends on belief about the agent's type or the realisation of the random variable which the agent can influence, the agent will exert greater effort to reduce this probability. The higher the chance that the relationship will be continued means the higher the expected return from the relationship.

The FOC (6.2) in Proposition 6.1 is a generalisation of the FOC (4.2) derived in Section 4.2 where product discontinuation policy is not considered. One can obtain condition (4.2) by setting m^* in (6.2) approaches $-\infty$ (where the product will never be discontinued). To

verify this, rewrite (6.2) as the following,

$$\begin{split} \sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(\lambda^* \right) \left[\mu_t' - \mu_t' \left(\left(\frac{\phi \left(\frac{m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t}} \right)}{1 - \Phi \left(\frac{m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t}} \right)} \right)^2 - \frac{\phi \left(\frac{m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t}} \right) m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t} \left(1 - \Phi \left(\frac{m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t}} \right) \right)} \right) - C' \left(q^* \right) \right] \\ + \sum_{t=1}^{\infty} \delta^{t-1} \left. \frac{\partial}{\partial q} \kappa_t \left(q, \lambda^* \right) \right|_{q=q^*} \left(q^* + \sigma_{m_t} \frac{\phi \left(\frac{m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t}} \right)}{1 - \Phi \left(\frac{m^* \left(\lambda^*, q^* \right)}{\sigma_{m_t}} \right)} - C \left(q^* \right) \right) = 0 \end{split}$$

since for the standard normal distribution, $\phi'(v) = -v\phi(v)$, the expression becomes

$$\sum_{t=1}^{\infty} \delta^{t-1} \kappa_t \left(\lambda^*\right) \left[\mu_t' - \mu_t' \left(\left(\frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} \right)^2 + \frac{\phi'\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{\sigma_{m_t}\left(1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)\right)} \right) - C'(q^*) \right] + \sum_{t=1}^{\infty} \delta^{t-1} \left. \frac{\partial}{\partial q} \kappa_t\left(q, \lambda^*\right) \right|_{q=q^*} \left(q^* + \sigma_{m_t} \frac{\phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)}{1 - \Phi\left(\frac{m^*(\lambda^*, q^*)}{\sigma_{m_t}}\right)} - C(q^*) \right) = 0 \quad (6.3)$$

as $m^* \to -\infty$, $\kappa_t(\lambda^*) \to 1$ or the product will never be discontinued and $\frac{\partial}{\partial q}\kappa_t(q,\lambda^*)\Big|_{q=q^*} \to 0$. Since $\phi\left(\frac{m^*(\lambda^*,q^*)}{\sigma_{m_t}}\right)$ and $\Phi\left(\frac{m^*(\lambda^*,q^*)}{\sigma_{m_t}}\right)$ approach 0 as $m^* \to -\infty$, the second term in LHS of (6.3) vanishes. and because the slope of standard normal pdf, $\phi'\left(\frac{m^*(\lambda^*,q^*)}{\sigma_{m_t}}\right)$, converges to 0 as $m^* \to -\infty$. Therefore, as $m^* \to -\infty$, equation (6.3) collapses to

$$\sum_{t=1}^{\infty} \delta^{t-1} \left[\mu'_t - C'(q^*) \right] = 0$$

but since $\mu'_t = \frac{\partial}{\partial q} \mu_t(q) = \frac{(t-1)h_{\varepsilon}}{(t-1)h_{\varepsilon}+h_1}$, this is identical to (4.2).

Note that, when the product is never discontinued $(m^* \to -\infty)$, the incentive to raise quality in order to increase the probability of product continuation disappears. Only the incentive to bias consumers' learning that increases the consumers' valuation and the firm's expected return remains; this is the standard career concerns mechanism.

7 Conclusion

This paper has considered and investigated quality choice and product discontinuation problems under a moral hazard problem where information about product quality is incomplete but symmetric; information about true product quality is unknown to both consumers and firm. The firm's intended quality choice is difficult to control and subject to random shocks. This shock can be interpreted as a match between consumer's preference and product characteristics. The true quality is unobservable, and consumers imperfectly report their (noisy) experiences from the product to the next generation of consumers. Reports serve as noisy signals of the true quality. The market forms beliefs about a product's true quality based on the observed sequence of noisy signals or a product's reputation.

The paper started by introducing a career concerns type of model and investigating relationships between optimal quality choice and various variables. It was found that a firm offers a better quality product when the product remains longer on the market (i.e. has a longer life), the firm cares about long term relationship (i.e. has a higher discount factor), the precision of initial prior belief is low, and the signal is very informative. The mechanism behind these results is the standard career concerns' incentive (Holmström 1982). According to this model, the firm has an incentive to signal jam the consumer's learning process since consumers cannot tell whether perceived quality is due to characteristics that match their preference (i.e. product is of better type) or just because of the imperfect communication of experiences (i.e. high realisations of experience or report shocks). High quality is intended to bias consumer belief that a product fulfils their preferences.

The paper then moved on to consider the case in which a firm can decide to discontinue a product before its true quality is completely revealed and then relaunch a new product activating a new market learning process. The scenario in which the product discontinuation period is predetermined or fixed in advance was considered in the first instance. Here it was shown that longer product life increases the expected return, and that it is better for a firm to have one product with an infinite life than many products with a shorter life. In the deterministic stopping period case, a firm therefore never opts for a predetermined discontinuation period.

The deterministic stopping period assumption was then relaxed. The case of a firm choosing to discontinue and relaunch a product whenever and as often as it wants was then considered. In this context, the firm can choose to discontinue the product after its reputation for that period is realised. The paper proved that optimal product discontinuation policies or strategies exist and the optimal discontinuation policy was then derived. Comparative static analysis has been carried out and demonstrated that when a firm cares about the long-term relationship, it applies a more stringent quality-control discontinuation policy and the product tends to be discontinued sooner almost surely. This is also true when the precision of initial prior belief is low. The informativeness of the signal has an ambiguous effect on the policy.

The optimal condition for quality with discontinuation policy was derived and has been shown to be a generalisation of the quality choice first order condition when discontinuation decision is not allowed. Under the presence of discontinuation policy, a quality choice chosen by the firm, again, misleads consumers that the product possesses favourable characteristics. However, as opposed to standard career concerns models without product discontinuation, this affects the expected return by two means. The first is through increased consumer willingness to pay for the product; the standard career concerns mechanism. And secondly, the higher quality will increase chances of surviving the discontinuation policy, enhance the probability of product continuation and raise firm's expected return.

When the firm relaunches the product, the firm's "type" is re-drawn independently from the distribution identical to all of the previous products. In this case, consumers cannot use information about the previous products evaluate the new product's true quality. However, in the real world consumers have diverse expectations about the quality of new products launched by different firms or brands believing them to possess different characteristics or types that are permanent or at least stable. It would be interesting for future research to incorporate into this model the adverse selection aspect when firms use quality to signal their types. Another interesting dimension would be the role of advertising that, in addition to quality choice, could confuse and mislead consumer inference. This would offer an insight into the firm's dilemma between enhancing quality and intensifying advertising.

References

- CABRAL, LUÍS M B. (2000), "Stretching Firm and Brand Reputation", RAND Journal of Economics, 31(4), 658-673.
- [2] CABRAL, LUÍS M B. (2005), The Economics of Trust and Reputation: A Primer, Mimeo.

- CHOW, YUAN SHIH AND HERBERT E. ROBBINS (1963), "On Optimal Stopping Rules", Zeitschrift für Wahrscheinlichkeitstheorie und verwandte Gebiete, 2, 33-49.
- [4] COLE, HAROLD L., JAMES DOW, AND WILLIAM B. ENGLISH (1995), "Default, Settlement, and Signalling: Lending Resumption in a Reputational Model of Sovereign Debt", International Economic Review, 36(2), 365-385.
- [5] DUTTA, PRAJIT K. AND ALDO RUSTICHINI (1993), "A Theory of Stopping Time Games with Applications to Product Innovations and Asset Sales", *Economic Theory*, 3, 743-763.
- [6] DYBVIG, PHILIP H. AND CHESTER S. SPATT (1982), "Does It Pay to Maintain a Reputation?" Mimeo.
- [7] FAMA, EUGENE F. (1980), "Agency Problems and the Theory of the Firm", Journal of Political Economy, 88(2), 288-307.
- [8] FENG, YOUYI AND GUILLERMO GALLEGO (1995), "Optimal Starting Times for Endof-Season Sales and Optimal Stopping Times for Promotional Fares", *Management Science*, 41(8), 1371-1391.
- [9] FERGUSON, THOMAS S. (2000), Optimal Stopping and Applications, Unpublished Manuscript, University of California, Los Angeles.
- [10] GITTINS, JOHN C. AND D. M. JONES (1974), "A Dynamic Allocation Index for the Sequential Design of Experiments" in *Progress in Statistics*. J. Gani, K. Sarkadi and I. Vincze, eds., European Meeting of Statisticians, 1972, 1, North Holland: Amsterdam, 241-266.
- [11] HOLMSTRÖM, BENGT (1982), "Managerial Incentive Problems: A Dynamic Perspective" in Eassay in Economics and Management in Honour of Lars Wahlbeck, pp. 209-230. Swedish School of Economics and Business Administration, Helsinki, published in Review of Economic Studies, 66(1), January 1999, 169-182.
- [12] HÖRNER, JOHANNES (2002), "Reputation and Competition", American Economic Review, 92(3), 644-663.

- [13] KATEHAKIS, MICHAEL N. AND ARTHUR F. VEINOTT, JR (1987), "The Multi-Armed Bandit Problem: Decomposition and Computation", *Mathematics of Operations Re*search, 12(2), 262-268.
- [14] KLIEN, BENJAMIN AND LEFFLER, KEITH B. (1981), "The Role of Market Forces in Assuring Contractual Performance", Journal of Political Economy, 89(4), 615-641.
- [15] KREP, DAVID M. (1990), "Corporate Culture and Economic Theory" in *Perspective on Positive Political Economy*. James E. Alt and Kenneth A. Shepsle, eds., Cambridge University Press: Cambridge, Massachusette, 90-143.
- [16] KREP, DAVID M., PAUL MILGROM, JOHN ROBERT AND ROBERT WILSON (1982), "Rational Cooperation in the Finitely Repeated Prisoners' Dilemma", *Journal of Economic Theory*, 27(2), 245-252.
- [17] KREP, DAVID M. AND ROBERT WILSON (1982), "Reputation and Imperfect Information", Journal of Economic Theory, 27(2), 253-279.
- [18] MAILATH, GEORGE J. AND LARRY SAMUELSON (1998), "Your Reputation Is Who You're Not, Not Who You'd Like to Be", CARESS Working Paper No.98-11, University of Pennsylvania.
- [19] MAILATH, GEORGE J. AND LARRY SAMUELSON (2001), "Who Wants a Good Reputation?", *Review of Economic Studies*, 68(2), 415-441.
- [20] MAILATH, GEORGE J. AND LARRY SAMUELSON (2006), Repeated Games and Reputations: Long-Run Relationships, Oxford University Press: New York.
- [21] MILGROM, PAUL AND JOHN ROBERTS (1982), "Predation, Reputation and Entry Deterrence" Journal of Economic Theory, 27(2), 280-312.
- [22] PHELAN, CHRISTOPHER. (2001), "Public Trust and Government Betrayal", Journal of Economic Theory, 130(1), 27-43.
- [23] ROGERSON, WILLIAM P. (1983), "Reputation and Product Quality", Bell Journal of Economics, 14(2), 508-516.
- [24] SHAPIRO, CARL (1983), "Premiums for High Quality Products as Returns to Reputations", Quarterly Journal of Economics, 98(4), 659-679.

- [25] SNELL, LAURIE J. (1952), "Application of Martingale System Theorems", Trans. Amer. Math. Soc, 73, 293-312.
- [26] TADELIS, STEVEN (1999), " "What's in a Name?" Reputation as a Tradable Asset", American Economic Review, 89(3), 548-563.
- [27] TADELIS, STEVEN (2002), "The Market for Reputations as an Incentive Mechanism", Journal of Political Economy, 92(2), 854-882.
- [28] VARAIYA, PRAVIN., JEAN WALRAND AND CAGATAY BUYUKKOC (1985), "Extensions of the Multiarmed Bandit Problem: The Discounted Case", *IEEE Trans. Autom. Control AC-30*, 426-439.
- [29] VON WEIZSÄCKER, CARL C. (1980), "A Welfare Analysis of Barriers to Entry", Bell Journal of Economics, 11(2), 399-420.
- [30] WHITTLE, PETER. (1980), "Multi-Armed Bandits and the Gittins Index", Journal of the Royal Statistical Society: Series B (Statistical Methodology), 42, 143-144.
- [31] WHITTLE, PETER. (1982), Optimization over Time. 1. John Wiley: New York, 210-220.